CENTRE for ECONOMIC

CEP Discussion Paper No 1553

June 2018

Managing Trade: Evidence from China and the US

Nicholas Bloom Kalina Manova Stephen Teng Sun John Van Reenen Zhihong Yu





Abstract

We present a heterogeneous-firm model in which management ability increases both production efficiency and product quality. Combining six micro-datasets on management practices, production and trade in Chinese and American firms, we find broad support for the model's predictions. First, better managed firms are more likely to export, sell more products to more destination countries, and earn higher export revenues and profits. Second, better managed exporters have higher prices, higher quality, and lower quality-adjusted prices. Finally, they also use a wider range of inputs, higher quality and more expensive inputs, and imported inputs from more advanced countries. The structural estimates indicate that management is important for improving product on efficiency and product quality in both countries, but it matters more in China than in the US, especially for product quality. Panel analysis for the US and a randomized control trial in India suggest that management exerts causal effects on product quality, production efficiency, and exports. Poor management practices may thus hinder trade and growth, especially in developing countries.

Key words: management, exports, product quality, productivity JEL: F10; F14; F23; L20; O19; O32

This paper was produced as part of the Centre's Growth Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

Any opinions and conclusions expressed herein are those of the authors and do not necessarily represent the views of the U.S. Census Bureau. All results have been reviewed to ensure that no confidential information is disclosed. We thank our formal discussants Pol Antras, David Atkin, Luis Garicano, Penny Goldberg, Elhanan Helpman, Amit Khandelwal, Veronica Rappoport, Stephen Redding, Eric Verhoogen, Stephen Yeaple, and Daniel Xu, and seminar and conference participants at AEA, Barcelona GSE, CEPR ERWIT, CEPR Workshop on Incentives, Management & Organization, EEA, Hitotsubashi Conference on International Trade & FDI, NBER- CCER China Conference, NBER ITI, Nottingham GEP, Princeton IES, RIEF, SCID/IGC Conference on Trade, Productivity & Development, Stockholm-CCER China Conference, Villars Research Workshop in International Trade, West Coast Trade Workshop; Aarhus, Birmingham, Cambridge, CityUHK, Geneva, HKU, HKUST, Kiel Institute for World Economy, KOF-ETH-UZH Zürich, LMU Munich, Louvain-la-Neuve, Luxembourg, MIT Sloan, National Bank of Belgium, Nottingham, Oslo, Oxford, Stanford, Sussex, UC Berkeley, UCL, VU Amsterdam, Warwick. We are grateful for funding from the ESRC, European Research Council (grant agreement 724880), National Science Foundation and Sloan Foundation.

Nicholas Bloom, Stanford University. Kalina Manova, University College London and Centre for Economic Performance, London School of Economics. Stephen Teng Sun, Peking University. John Van Reenen, MIT and Centre for Economic Performance, London School of Economics. Zhihong Yu, University of Nottingham.

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

© N. Bloom, K. Manova, S.T. Sun, J. Van Reenen and Z. Yu, submitted 2018.

1 Introduction

Productivity, management practices and international trade activity vary dramatically across firms and countries (Bernard, Jensen, Redding, and Schott, 2007; Syverson, 2011). In the literature, higher measured Total Factor Productivity (TFP) has been associated with export success and superior management with higher profits. However, measured TFP is subject to estimation biases and constitutes a residual "black box", while the mechanisms through which management operates remain largely unknown. From a policy perspective, improving firm capabilities is important for stimulating firm performance and aggregate growth, but this requires knowledge of the determinants of firm productivity. While it is widely believed that management strategies play a central role, especially in emerging economies trying to move up the quality ladder (Sutton, 2012), the scant evidence for this is primarily from case studies.

In this paper we perform what we believe is the first large-scale analysis of the role of management practices for export performance and in the process shed light on these questions. We uncover novel empirical facts and interpret them through the lens of a heterogeneous-firm theoretical model that disciplines the estimation approach. We study the world's two largest export economies - China and the United States - and find consistent empirical patterns in both countries despite their very different income levels, institutional quality and market frictions. In particular, we exploit unique new data on plant-level production, plant-level management practices, and transaction-level international trade activity for 485 Chinese firms in 1999-2008 and over 10,000 US firms in 2010.

We first establish that better managed firms have superior export performance. Companies with more effective management practices are systematically more likely to engage in exporting. Conditional on exporting, they sell more products to more destination countries and earn higher export revenues and profits. In addition, our findings for management survive when we explicitly control for revenue-based firm TFP as commonly constructed in the literature.

We then present a collection of independent results that jointly inform the mechanisms through which management strategies affect firm performance. On the sales side, better managed firms charge higher export prices within narrowly defined destination-product markets. We estimate a model-consistent indicator of product quality, and show that management competence is associated with higher output quality and lower quality-adjusted prices. On the production side, better managed companies use more expensive, higher-quality imported inputs and more inputs from suppliers located in developed economies. They also source a wider range of distinct inputs from more countries of origin.

We propose that these empirical patterns are consistent with management competence being an important component of firms' total factor productivity, whereby more effective managerial practices increase both production efficiency and quality capacity. Superior management enables firms to use more sophisticated, higher-quality inputs and more complex assembly technologies that increase output quality. At the same time, advanced management allows firms to process inputs and execute assembly more cheaply. When both the efficiency and quality channels are active, they push marginal costs in opposite directions, such that the net effect of management competence on prices and quantities is ambiguous, but it unambiguously raises product quality, sales and profits. These predictions are preserved when we extend the baseline model to incorporate endogenous input choice, endogenous management practices, or non-management components of TFP.

Our main empirical analysis explores the cross-sectional variation in management and trade activity across Chinese and American firms. We therefore do not distinguish between a causal effect of good management and an equilibrium relationship between joint outcomes of the firm's profit maximization problem. Instead, we view our baseline results as conditional correlations that inform the mechanisms through which management operates. We also provide two additional pieces of evidence that suggest a causal role for management competence. These are based on panel analysis of changes within US firms over time and on a randomized control trial that offered management consulting to Indian firms. Both exercises reveal patterns closely in line with the cross-section.

Our findings address two open questions in two separate, active literatures. A large theoretical and empirical literature in international trade emphasizes the role of firm productivity as a key determinant of firms' export performance (Melitz, 2003; Bernard, Eaton, Jensen, and Kortum, 2003; Melitz and Ottaviano, 2008; Bernard, Jensen, Redding, and Schott, 2007). More productive firms have been found to export more products to more destinations, thereby generating higher export revenues and profits. This body of work conceptualizes firm productivity as TFPQ, or the ability to manufacture at low marginal costs, such that more productive firms are more successful exporters because they set lower prices. Recent analyses point to the importance of product quality as well, showing that more successful exporters use higher-quality manufactured inputs and more skilled workers in order to produce higher-quality output that sells at higher prices (Verhoogen, 2008; Kugler and Verhoogen, 2012; Khandelwal, 2010; Manova and Zhang, 2012; Johnson, 2012; Bastos, Silva, and Verhoogen, 2018). Yet productivity is typically measured as TFPR, or a revenue-based residual from production function estimates. This makes it subject to estimation bias and complicates the interpretation of trade-TFPR regression analyses (Ackerberg, Caves, and Frazer, 2015; De Loecker, 2011). Thus, an important open question in the trade literature is what constitutes productivity, how it should be measured, and what explains its vast variation across firms.

A separate and older literature has examined the relationship between firm management, productivity and performance (Walker, 1887; Taylor, 1911; Syverson, 2011). One likely route for this management-productivity link emphasized by the management literature is through lean manufacturing and improved quality (Deming, 1950; Womack, Jones, and Roos, 1990; Drew, McCallum, and Roggenhofer, 2016; Sutton, 2007). Yet there is no systematic, direct evidence

on the mechanisms through which management operates.¹

Our work informs both of these open questions. We conclude that effective management enhances firm performance by enabling firms to manufacture higher-quality goods more efficiently, such that both production efficiency and quality capability increase with management competence. We also unpack the black box of TFPR and identify management practices as a concrete, tangible and directly measured TFPQ component that accounts for the heterogeneity in firm performance. Studying management practices thus circumvents concerns with estimation biases in trade-productivity regressions and delivers clearer policy lessons.

This paper also adds to recent research on the impact of trade liberalization on the organization of production inside firms. Evidence indicates that trade reforms incentivize firms to change the number of management layers, adjust the number and wages of managers and workers along the occupational hierarchy, and upgrade management practices (Caliendo and Rossi-Hansberg, 2012; Caliendo, Monte, and Rossi-Hansberg, 2017; Chakraborty and Raveh, 2018; Chen and Steinwender, 2016). At the same time, improved access to imported inputs is important to the product quality, product scope and export success of firms in developing countries, because of the limited domestic supply of high-quality specialized inputs and equipment (Goldberg, Khandelwal, Pavcnik, and Topalova, 2010; Fieler, Eslava, and Xu, 2018; Manova and Zhang, 2012). This matters since poor economies often rely on international trade for growth, and specifically on exporting to large, developed and profitable markets that maintain high quality standards. In light of these two lines of work, our results suggest that poor managerial practices may impede trade, growth and entrepreneurship in the world's poorest economies.

Finally, our findings speak to the active literature on the implications of firm heterogeneity for aggregate productivity, welfare and the gains from trade (Hsieh and Klenow, 2009; Arkolakis, Costinot, and Rodríguez-Clare, 2012; Melitz and Redding, 2013). Evidence indicates that reallocations across firms and across products within firms, as well as productivity upgrading within firms contribute significantly to the aggregate adjustment to trade reforms and macroeconomic shocks (Pavcnik, 2002; Bernard, Eaton, Jensen, and Kortum, 2003; Bustos, 2011; Berthou, Chung, Manova, and Bragard, 2016). Understanding the sources of firm heterogeneity is thus important for understanding aggregate outcomes. In addition, given evidence on the complementarity between manufactured input quality and skilled labor in the production of output quality (Verhoogen, 2008), quality differentiation across firms and its interplay with management competence also has implications for the differential effects of shocks across the firm size and worker skill distributions.

The remainder of the paper is organized as follows. Section 2 theoretically models the role of management competence for firms' export performance. Section 3 introduces the Chinese and

¹The most popular management systems - Six-Sigma, Lean, and the Toyota Production System - all emphasize that improving productivity and quality is best achieved by an ongoing focus on reducing defects. In fact, this approach is now so popular that it has spread from manufacturing across most sectors, for example to Lean Retail (Myerson, 2014), Lean Healthcare (Group, 2014) and even Lean Government (Teeuwen, 2010).

US data on firms' balance sheets, trade activity, and management practices. Section 4 examines the relationship between firms' management competence and export performance, and explores how managerial strategies contribute to firm productivity. Section 5 analyzes the efficiency and quality mechanisms through which superior management operates. Section 6 provides evidence from both a randomized control trial and within-firm changes in the panel that is consistent with causal effects of good management. The last section concludes.

2 Theoretical Framework

We develop a theoretical model of international trade in which heterogeneous firms choose how many products to manufacture, what markets to enter, and which products to sell in each market. In the baseline set-up, firms receive an exogenous draw of management competence which uniquely determines firm choices and performance outcomes. We consider the endogenous adoption of management practices in an extension to this benchmark model in Appendix 2.1. We posit that effective management can enhance firm performance by increasing production efficiency and/or quality capacity. We characterize the relationship between firms' management competence and trade activity under alternative assumptions about the relative importance of these two channels, and derive testable predictions that allow us to empirically assess their relevance in practice. We relegate all detailed proofs to Appendix 1.

We incorporate management competence in a partial-equilibrium trade model that features quality and efficiency differentiation across firms and across products within multi-product firms. In our baseline, we treat management effectiveness as equivalent to TFP, such that our model closely resembles that in Bernard, Redding, and Schott (2010)(BRS), Kugler and Verhoogen (2012), and Manova and Yu (2017). We examine the alternative in which management practices are one of multiple components of firm productivity in Appendix 2.2.

2.1 Set Up

Consider a world with J + 1 countries. In each country, a continuum of heterogeneous firms produce horizontally and vertically differentiated goods which they sell at home and potentially export abroad. Consumers exhibit love of variety such that the representative consumer in country j has CES utility $U_j = \left[\int_{i\in\Omega_j} (q_{ji}x_{ji})^{\alpha} di\right]^{\frac{1}{\alpha}}$, where q_{ji} and x_{ji} are the quality and quantity consumed by country j of variety i, and Ω_j is the set of goods available to j. The elasticity of substitution across products is $\sigma \equiv 1/(1-\alpha) > 1$ with $0 < \alpha < 1$. If total expenditure in country j is R_j , j's demand for variety i is $x_{ji} = R_j P_j^{\sigma-1} q_{ji}^{\sigma-1} p_{ji}^{-\sigma}$, where $P_j = \left[\int_{i\in\Omega_j} \left(\frac{p_{ji}}{q_{ji}}\right)^{1-\sigma} di\right]^{\frac{1}{1-\sigma}}$ is a quality-adjusted ideal price index and p_{ji} is the price of variety iin country j. Quality is thus defined as any objective attribute, subjective taste preference or other demand shock that increases the consumer appeal of a product given its price. Note that a sufficient statistic for unobserved product quality $\ln q_{ji}$ within market j can be constructed from observed price and quantity data as $\sigma \ln p_{ji} + \ln x_{ji}$, since R_j and P_j do not vary across products sold in j (Khandelwal, 2010; Khandelwal, Schott, and Wei, 2013).

2.2 Production and Sales Technology

The production technology is characterized by a production function for physical units of output and a production function for output quality. Firms' management competence can affect both the ability to assemble given inputs at low cost and the capacity to make high-quality goods. We refer to these two mechanisms as *production efficiency* and *quality capacity*.

In order to begin manufacturing, entrepreneurs have to incur sunk entry costs associated with research and product development. They face uncertainty about their production efficiency and product quality, and observe them only after completing this irreversible investment. At that point they decide whether to exit immediately or commence production and possibly export.

Upon entry, firms draw firm-wide managerial ability $\varphi \in (0, \infty)$ from distribution $g(\varphi)$ and a vector of firm-product specific expertise levels $\lambda_i \in (0, \infty)$ from distribution $z(\lambda)$. We will think of better managed firms as having a higher ability draw φ .^{2,3} Since the success of research and product development may differ across products within a firm, we assume that $g(\varphi)$ and $z(\lambda)$ are independent of each other and common across firms with continuous cumulative distribution functions $G(\varphi)$ and $Z(\lambda)$ respectively, while λ is i.i.d. across products and firms.

Producing one unit of physical output requires $(\varphi \lambda_i)^{-\delta}$ units of labor whose wage is normalized to 1 to serve as the numeraire. The parameter $\delta > 0$ governs the extent to which good management practices can lower unit input requirements and increase the efficiency with which these inputs are assembled into final goods. Intuitively, effective management can improve production efficiency by optimizing inventory control, synchronizing and monitoring production targets across manufacturing stages, reducing wastage, incentivizing workers, etc.

At a marginal cost of $(\varphi \lambda_i)^{\theta-\delta}$ workers, the firm produces one unit of product *i* with quality $q_i (\varphi, \lambda_i) = (\varphi \lambda_i)^{\theta}, \theta > 0$. This reduced-form quality production function captures the idea that manufacturing goods of higher quality is associated with higher marginal costs because it requires the use of more complex assembly processes and more expensive intermediate inputs of higher quality. For example, while sewing a dress using unskilled labor, cotton and plastic buttons might entail the same assembly process as sewing a dress using skilled labor, silk and mother-of-pearl buttons, the latter utilizes more expensive inputs and is considered higher quality. Similarly,

²As we show in Appendix 2.1, one can endogenize management practices while retaining the key features of the model. For example, entrepreneurs might receive an exogenous talent draw ϕ , choose management practice $m(\phi)$, and face marginal costs and quality that depend on $\varphi = \phi m(\phi) \lambda_i$. If the cost f_m of adopting m satisfies $df_m/dm > 0$ and $d^2 f_m/dm^2 > 0$, then Propositions 1-4 hold both for effective firm ability φ and for management $m(\phi)$.

³We show in Appendix 2.2 that our theoretical predictions would continue to hold if management is one of multiple draws that jointly determine firm ability. For example, firm ability $\varphi = m \cdot \phi$ may depend on the entrepreneur's intrinsic talent ϕ and the manager's competence for implementing effective management practices m. If entrepreneurs and managers do not match perfectly assortatively due to labor market frictions, then $|corr(m, \phi)| \neq 1$. While all firm outcomes would now be pinned down by φ instead of m alone, management competence would have the same effects as in our baseline model *ceteris paribus*.

while a printer built from 50 components might only be able to print, the sophisticated assembly of 150 parts might produce a multi-functional printer that can print, scan and photocopy. The parameter θ reflects the degree to which superior management enables firms to produce higherquality products. Intuitively, effective management can enhance quality capacity by tightening quality control, facilitating specialized assembly, minimizing costly mistakes, etc.

For expositional simplicity, we do not explicitly model firms' input choice in our baseline set-up, but follow Baldwin and Harrigan (2011) in assuming that product quality is fixed by exogenous draws. In Appendix 2.3, we formally establish that endogenizing input quality in a richer framework would preserve our theoretical predictions. Following Kugler and Verhoogen (2012), we show how complementarity between firm ability and input quality in the production function for output quality would induce more capable firms to use higher-quality inputs in order to produce higher-quality goods.

Firms' marginal cost thus reflects two opposing forces: On the one hand, better managed firms have higher production efficiency. On the other hand, better managed firms produce higher quality using more expensive inputs and/or more complex assembly. The net effect of these two forces on marginal costs is theoretically ambiguous and depends on the relative magnitudes of θ and δ .

We make a number of standard assumptions about firms' production and sales costs that are motivated by salient patterns in the data. Firms incur a fixed operation cost of headquarter services f_h and a fixed overhead cost f_p for each active product line, in units of labor. This will imply that companies with different ability draws will choose to produce a different number of products. Entering each foreign market j is associated with additional headquarter services f_{hj} necessary for complying with customs and other regulations, as well as for the maintenance of distribution networks. As a result, some low-ability sellers in the domestic market will not become exporters or will supply some but not all countries. Finally, exporting entails destination-product specific fixed costs f_{pj} (constant across products within j, but varying across countries), which reflect market research, product customization and standardization, and advertising. There are also variable transportation costs such that τ_j units of a good need to be shipped for 1 unit to arrive. These trade costs will ensure that firms might not offer every product they sell at home in every foreign market they enter.

2.3 Profit Maximization

Firms must decide which products to produce, where to sell them and at what prices in order to maximize profits from their global operations. With monopolistic competition and a continuum of varieties, individual producers take all aggregate expenditures R_j and price indices P_j as given, and separately maximize profits in each country-product market.⁴ A firm with management competence φ will choose the sales price and quantity of a product with expertise draw λ_i in

 $^{^{4}}$ See Eckel, Iacovone, Javorcik, and Neary (2015) and Eckel and Neary (2010) for an alternative model which incorporates product cannibalization effects.

country j by solving

$$\max_{p_{ji}, x_{ji}} \pi_{ji} (\varphi, \lambda_i) = p_{ji} (\varphi, \lambda_i) x_{ji} (\varphi, \lambda_i) - \tau_j x_{ji} (\varphi, \lambda_i) (\varphi \lambda_i)^{\theta - \delta} - f_{pj}$$
(1)
s.t. $x_{ji} (\varphi, \lambda_i) = R_j P_j^{\sigma - 1} q_{ji} (\varphi, \lambda_i)^{\sigma - 1} p_{ji} (\varphi, \lambda_i)^{-\sigma}.$

Producers therefore charge a constant mark-up $\frac{1}{\alpha}$ over marginal cost, and have the following price, quantity, quality, quality-adjusted price, revenues and profits for product *i* in market *j*:

$$p_{ji}(\varphi,\lambda_i) = \frac{\tau_j(\varphi\lambda_i)^{\theta-\delta}}{\alpha}, \qquad x_{ji}(\varphi,\lambda_i) = R_j P_j^{\sigma-1} \left(\frac{\alpha}{\tau_j}\right)^{\sigma} (\varphi\lambda_i)^{\delta\sigma-\theta}, \tag{2}$$

$$q_i(\varphi,\lambda_i) = (\varphi\lambda_i)^{\theta}, \quad p_{ji}(\varphi,\lambda_i)/q_i(\varphi,\lambda_i) = \frac{\tau_j(\varphi\lambda_i)^{-\theta}}{\alpha}, \quad (3)$$

$$r_{ji}(\varphi,\lambda_i) = R_j \left(\frac{P_j\alpha}{\tau_j}\right)^{\sigma-1} (\varphi\lambda_i)^{\delta(\sigma-1)}, \qquad \pi_{ji}(\varphi,\lambda_i) = \frac{r_{ji}(\varphi,\lambda_i)}{\sigma} - f_{pj}.$$
(4)

When j corresponds to the firm's home market, there are no iceberg costs ($\tau_j = 1$) and the destination-product fixed cost f_{pj} is replaced by the product-specific overhead cost f_p . Note that the empirical analysis examines free-on-board export prices and revenues, that is $p_{ji}^{fob}(\varphi, \lambda_i) = \frac{(\varphi \lambda_i)^{\theta-\delta}}{\alpha}$ and $r_{ji}^{fob}(\varphi, \lambda_i) = R_j (P_j \alpha)^{\sigma-1} (\varphi \lambda_i)^{\delta(\sigma-1)}$.

If $\theta = 0$ and $\delta > 0$, effective management improves firm performance only by increasing production efficiency but the quality channel is moot. The model then reduces to the BRS framework in which all firms offer the same product quality level, but better managed firms have lower marginal costs and therefore set lower prices, sell higher quantities, and earn higher revenues and profits. While formally $\delta = 1$ in BRS, this normalization is immaterial when $\theta = 0$.

Conversely, if $\theta > 0$ and $\delta = 0$, management competence benefits firm performance by improving product quality but the production efficiency mechanism is not active. Now all firms share the same quality-adjusted prices, revenues and profits, but better managed companies charge higher prices, offer higher quality and sell lower quantities.

The most interesting scenario arises when $\theta > 0$ and $\delta > 0$, such that management operates through both the production efficiency and the product quality channels. We focus on this scenario below as it is most relevant empirically. In this case, superior management is unambiguously associated with higher product quality, lower quality-adjusted prices, higher revenues and higher profits. However, the implications for quantity and price levels are theoretically ambiguous. If $\theta > \delta$, as management competence grows, product quality rises sufficiently quickly with the cost of sophisticated inputs and assembly to overturn the effects of improved production efficiency. As a result, effective management corresponds to *higher* output prices. If $\theta < \delta$ by contrast, good management practices translate into *lower* prices. In the knife-edge case of $\theta = \delta$, production efficiency and product quality are equally elastic in management capacity, and prices are invariant across the firm management distribution. Finally, better managed firms sell higher quantities if and only if $\sigma \delta > \theta$.

2.4 Selection into Products and Markets

Consumers' love of variety and the presence of product-specific overhead costs f_p imply that no firm will export a product without also selling it at home. In turn, firms optimally manufacture only goods for which they can earn non-negative profits domestically. Since profits increase with product expertise λ_i , there is a zero-profit expertise level $\lambda^*(\varphi)$ for each management ability draw φ below which firm φ will not make *i*. This value is defined by:

$$\pi_d(\varphi, \lambda^*(\varphi)) = 0 \Leftrightarrow r_d(\varphi, \lambda^*(\varphi)) = \sigma f_p, \tag{5}$$

where d indicates that revenues are calculated for the domestic market.

Recall that product expertise is independently and identically distributed across goods. By the law of large numbers, the measure of varieties that a firm with ability φ produces equals the probability of an expertise draw above $\lambda^*(\varphi)$, or $[1 - Z(\lambda^*(\varphi))]$. Since $d\lambda^*(\varphi)/d\varphi < 0$, better managed firms have a lower zero-profit expertise cut-off and offer more products. One interpretation of this result is that better managed firms bring superior quality control to any product line. This can partially offset using less skilled workers or inputs of lower quality such that output quality and consumer appeal remain high.

Following the same logic, a firm with ability φ will export product *i* to country *j* only if its expertise draw is no lower than $\lambda_j^*(\varphi)$ given by:

$$\pi_{ji}\left(\varphi,\lambda_{j}^{*}\left(\varphi\right)\right) = 0 \Leftrightarrow r_{j}\left(\varphi,\lambda_{j}^{*}\left(\varphi\right)\right) = \sigma f_{pj}.$$
(6)

The measure of products that firm φ sells to j is thus $\left[1 - Z\left(\lambda_j^*(\varphi)\right)\right]$. Since $d\lambda_j^*(\varphi)/d\varphi < 0$, better managed firms export more products than worse run firms to any given destination.

When the exporting expertise cut-off lies above the zero-profit expertise cut-off, $\lambda_j^*(\varphi) > \lambda^*(\varphi)$, there will be selection into exporting. Across products within a firm, not all goods sold at home will be shipped to j. Similarly, across firms supplying a product domestically, not all will be able to market it abroad. Given the overwhelming evidence for both patterns in the prior literature, we assume that $\lambda_j^*(\varphi) > \lambda^*(\varphi)$ holds for all j.

For every management level φ , the expertise cut-off for exporting generally varies across destinations because the market size R_j , price index P_j , variable τ_j and fixed f_{pj} trade costs are country specific. Firms therefore adjust their product range across markets. Each exporter follows a product hierarchy and adds goods in decreasing order of expertise until it reaches the marginal product that brings zero profits. Within a supplier, higher-quality goods are shipped to more countries, earn higher revenues in any given market, and generate higher worldwide sales.

Firms enter a market only if total expected revenues there exceed all associated costs. The export profits in country j of a firm with management competence φ are:

$$\pi_j(\varphi) = \int_{\lambda_j^*(\varphi)}^{\infty} \pi_j(\varphi, \lambda) \, z\left(\lambda\right) d\lambda - f_{hj}.$$
(7)

Export profits $\pi_j(\varphi)$ increase with management ability because better managed firms sell more products in j (i.e. lower $\lambda_j^*(\varphi)$) and earn higher revenues from each good (i.e. higher $\pi_j(\varphi, \lambda)$) than firms with the same product expertise draws but worse management. Therefore only firms with management level above a cut-off φ_j^* will service destination j, where φ_j^* satisfies:

$$\pi_j\left(\varphi_j^*\right) = 0. \tag{8}$$

With asymmetric countries, φ_j^* varies across destinations and better managed firms enter more markets because they are above the export ability cut-off for more countries. Better managed exporters thus outperform worse run producers along all three margins: number of export destinations, product scope in each destination, and sales in each destination-product market.

Finally, not all firms that incur the sunk cost of entry survive. Once they observe their management ability and expertise draws, firms begin production only if their expected profits from all domestic and foreign operations are non-negative. Firm φ 's global profits are given by:

$$\pi\left(\varphi\right) = \int_{\lambda^{*}(\varphi)}^{\infty} \pi_{d}\left(\varphi,\lambda\right) z\left(\lambda\right) d\lambda + \sum_{j} \left(\int_{\lambda_{j}^{*}(\varphi)}^{\infty} \pi_{j}\left(\varphi,\lambda\right) z\left(\lambda\right) d\lambda - f_{hj}\right) - f_{h}.$$
(9)

The first integral in this expression captures the firm's domestic profits from all products above its expertise cut-off for production $\lambda^*(\varphi)$, while the summation represents worldwide export profits from all traded products and destinations.

Total profits increase in φ because better managed firms sell more products domestically, earn higher domestic revenues for each product, and have superior export performance as described above. Companies below a minimum management level φ^* are thus unable to break even and exit immediately upon learning their attributes. This cut-off is defined by the zero-profit condition:

$$\pi\left(\varphi^*\right) = 0. \tag{10}$$

2.5 Empirical Predictions

We summarize the key empirical predictions of the model with the following propositions. We take these predictions to the data, which we turn to next.

Proposition 1 Better managed firms are more likely to export.

Proposition 2 Better managed firms export more products to more destination markets and earn higher export revenues and profits.

Proposition 3 Better managed firms offer higher-quality products if $\theta > 0$, but quality is invariant across firms if $\theta = 0$. Better managed firms set lower quality-adjusted prices if $\delta > 0$, but quality-adjusted prices are invariant across firms if $\delta = 0$. Better managed firms charge higher prices if $\theta > \delta$ and lower prices if $\delta > \theta$, but prices are invariant across firms if $\theta = \delta$.

Proposition 4 Better managed firms use more expensive inputs of higher quality and/or more expensive assembly of higher complexity if $\theta > 0$, but input quality and assembly complexity are invariant across firms if $\theta = 0$.

3 Data

Our analysis makes use of unique, matched establishment- or firm-level data for the world's two largest exporters - China and the US - on production (ASIE and ASM), international trade (CCTS and LFTTD), and management practices (WMS and MOPS) respectively. We exploit six proprietary micro-data sources in total, three for each country, to assemble a dataset that is unprecedented in its coverage and detail. This section describes how management practices are evaluated, introduces the data, and summarizes key features of firm activity.

3.1 Measuring Management Practices

Systematic data on firms' management practices have only recently become available. Since the first major wave in 2004, the World Management Survey (WMS) has developed standardized measures of management competence for over 20,000 manufacturing firms located in 34 countries. It considers multiple aspects of firm management, and evaluates the relative effectiveness of different practices within each aspect. WMS is conducted via double-blind phone interviews with plant managers, and covers representative firm samples in a large number of countries.⁵ The Management and Organizational Practices Survey (MOPS), which was introduced by the US Census in 2010, is modeled after WMS and provides management scores for around 32,000 US manufacturing establishments.

WMS (MOPS) includes 18 (16) questions about the management of physical capital (monitoring and targets) and human resources (incentives) inside a firm, examples of which appear in Figure 1. A first set of questions pertain to the monitoring of progress towards production targets via the frequent collection, analysis and dissemination of multiple performance metrics. A second set of questions characterize the design, integration and realism of production targets. These questions assess to what extent targets are consistently set across production stages and tightly connected to performance, both in the short-run and long-run, for managers and nonmanagers. A final set of questions capture the use of incentives mechanisms to identify, promote and reward high performers with bonuses, while sanctioning underperformers.

Each management question is scored on a scale of 1 to 5 in WMS and 0 to 1 in MOPS, with higher values indicating more structured management involving greater monitoring, more aggressive targets and stronger performance incentives. For each country, we first standardize the responses to each question across all firms to be mean 0 and standard deviation 1. We then average across questions to obtain a single *management score* for each firm in order to be

⁵See Bloom and Van Reenen (2007) for full details of the survey process.

comprehensive. Finally, we standardize these management scores across firms in each country to be mean 0 and have standard deviation 1.

WMS and MOPS are based on the lean manufacturing and modern human resource management practices, as used by leading international firms of management consultants, to focus on core management practices that should benefit firm performance regardless of the industry or economic environment. Our analysis will nevertheless account for the possibility that the relevance of specific management practices might vary across industries with industry fixed effects. We also conduct all estimations separately for China and the US. This addresses potential concerns that the effectiveness of certain management practices might depend on the formal and informal institutions in a country (e.g. labor market flexibility, cultural norms, respect for managerial hierarchy). To the extent that the management surveys are biased towards successful production practices in the West, measurement error would introduce downward bias and work against us finding consistent patterns for both China and the US. In practice, evidence indicates that the introduction of more effective management practices according to WMS criteria significantly improved firm performance in a randomized control trial in India (see Section 6 below).

3.2 United States

We employ three comprehensive datasets on the activities of US firms. First, US management is assessed using MOPS, the first and only comprehensive management dataset of its kind. Introduced as a mandatory part of the US Census's 2010 Annual Survey of Manufacturing, it documents the management practices of about 32,000 manufacturing establishments in 2010 and 2005 (as a recall). The sample captures 5.6 million employees, which is more than half of US manufacturing employment. The distribution of the management score across plants is plotted in Figure 2A. MOPS also includes several variables that we use to control for potential noise in the management score, namely an indicator for filing census forms online, the tenure and seniority of the respondent, and the discrepancy between employment data reported in MOPS and ASM.

Second, we obtain standard balance-sheet data on these establishments from the US Annual Survey of Manufacturers (ASM), available from 1973-2012⁶. The ASM records the total output, value added, profits and production inputs (such as employment, capital expenditures, energy inputs and materials purchases) for about 45,000 plants that correspond to over 10,000 firms. We also observe firms' age, location (out of 50 states), and primary industry of activity in the US NAICS 6-digit industry classification. We measure the skill intensity of firms' production technology with the log average wage and the share of workers with a college degree, and firms' capital intensity with log net fixed assets per worker. We construct two proxies for firm productivity, log value added per worker and the revenue-based TFP residual from production function

⁶The 2010 MOPS was part of the 2009-2013 ASM panel, so all establishments in the MOPS were surveyed every year over this period. In prior years establishments are surveyed in years ending with a "2" or "7" as part of the Economic Census, and otherwise if they are part of that year's ASM panel. Since the ASM panel over-samples larger establishments, it tends to include a large fraction of export activity.

regressions à la Levinsohn-Petrin performed separately for each NAICS-6 industry.

Third, we use the US Longitudinal Federal Trade Transaction Database (LFTTD), which contains detailed information about the universe of US international trade transactions in 1992-2012, at over 100 million transactions a year. LFTTD reports the value, quantity, unit (e.g. dozens, kilograms, etc.) and organization (intra-firm vs. arm's length) of all firm-level exports (free on board) and all firm-level imports (cost, insurance and freight included) by country and product for around 7,000 different products in the 10-digit Harmonized System and around 5,000 product categories at the HS 8-digit level. The raw data enables us to compute transaction-level unit values to proxy prices. To ensure comparability, whenever we study product-level data, a product is defined by both its HS code and unit. Given the lumpiness and seasonality of international trade, we analyze annual trade flows at various levels of aggregation such as the firm, firm-product, firm-destination, and firm-product-destination.

We link ASM, LFTTD and MOPS using firms' tax identifier that is common to all three datasets.⁷ We perform our baseline analysis for the resultant cross-section of about 32,000 US establishments in 2010 with contemporaneous production, trade and management data. The firms in this matched sample are on average bigger and better performing than firms without management data, but appear representative in that the relationship between standard productivity, size and performance metrics is the same in both subsamples. We analyze recall MOPS data for 2005 and panel ASM and LFTTD data in robustness checks.

3.3 China

We exploit three comprehensive datasets on the activities of Chinese firms that closely mirror those for the US. First, we observe the management practices of 507 Chinese firms collected in 2006-2007 as part of the World Management Survey (WMS). The distribution of the management score across firms is plotted in Figure 2B. Unlike MOPS, WMS is run as a telephone survey, relying on endorsements by respected institutions and highly-trained interviewers (e.g. MBAs) to achieve a response rate of 45% in China. Firms with 100 to 5000 employees are sampled randomly, and interviews employ double-blind techniques to obtain unbiased responses from plant managers. WMS also gathers additional firm and interview demographics. Of these, we use information on firms' primary industry of affiliation (out of 82 SIC 3-digit industries) and a set of controls for potential survey noise (interviews' duration, day of week and time of day; interviewer ID; interviewee gender, reliability and competence as perceived by the interviewer).

Second, we access production data at the firm level for the 1999-2007 period from China's Annual Survey of Industrial Enterprises (ASIE). ASIE is collected by the National Bureau of Statistics and provides standard balance-sheet information for all state-owned firms and all

⁷For firms with multiple establishments in ASM, we aggregate up to the firm level by summing production variables across all establishments belonging to the same firm. In the case of MOPS, we take the employment-weighted average management score across plants within a firm, but our results are robust to using the simple average instead. We use the age, location and primary industry of activity of the firm headquarters.

private firms with sales above 5 million Chinese Yuan, for over 200,000 firms a year. In addition to output, profits, value added and production inputs, we also observe firms' age, ownership structure (private domestic, state-owned domestic, foreign-owned), location (out of 31 provinces), and primary industry of activity in the Chinese SIC 3-digit industry classification.

Third, we utilize comprehensive data on the universe of Chinese firms' cross-border transactions in 2000-2008 from the Chinese Customs Trade Statistics (CCTS), spanning over 100 million transactions a year. CCTS is collected by the Chinese Customs Office and reports the value and quantity of firm exports (free on board) and imports (cost, insurance and freight included) in U.S. dollars by product and trade partner for 243 destination/source countries and about 7,500 products in the 8-digit Harmonized System.⁸ We calculate unit values as the ratio of shipment values and quantities, and analyze trade flows at different levels of aggregation. While CCTS does not distinguish between arm's-length and intra-firm transactions, it indicates the trade regime under which each export and import flow occurs. China recognizes a formal processing trade regime that permits duty-free imports of inputs for further processing, assembly and re-exporting on behalf of a foreign buyer. Each trade transaction is thus labeled as ordinary or processing trade, and firms can and do legally engage in both operation modes.

Of the 507 Chinese firms in WMS, we are able to match 485 to ASIE using the unique firm identifier that is common to both databases. We obtain the complete ASIE record for these 485 firms during 1999-2007, which produces an unbalanced panel of 3,233 firm-year observations.

Since CCTS maintains an independent system of firm registration codes, it cannot be mapped directly into ASIE or WMS. We follow standard practice in the literature and match CCTS to ASIE using an algorithm based on firms' name, address and phone number. Using ASIE as a bridge, we match 296 companies from WMS to CCTS. We then match 58 of the remaining unmatched companies in WMS directly to CCTS firms by postcode and translated Chinese-to-English company names. We ensure match quality by manually researching company webpages and reports, etc. With this two-step matching procedure, we locate detailed CCTS trade data for 354 of the 507 WMS companies, for a match rate of 70%. Of these 354 firms, 11% only export, 17% only import, and 72% both export and import according to CCTS. This is consistent with the fact that about 60% of the matched WMS-ASIE firms report positive exports on their balance sheets, while more firms may appear in the comprehensive CCTS records.

3.4 Summary Statistics

As a first glance at the data, we summarize the substantial variation in management practices, production and trade activity across firms in China and the US in Appendix Table 1. Starting with the US, 45% of the 32,000 US establishments in our 2010 matched sample export. The

⁸While the US and China both adhere to a standardized international HS 6-digit product classification system, countries are free to record their trade activity at finer levels of disaggregation that are not readily comparable across nations. Our baseline analysis exploits the granularity of the US and Chinese customs data at the HS-8 digit level, but our results are robust to using aggregated trade flows at the common HS-6 digit level or the most disaggregated data for the US at the HS-10 digit level.

typical exporter sells 19 different HS-8 digit products to 13 destinations and, conditional on using imported inputs, imports 20 distinct products from 6 countries, with large dispersion around these means.⁹ These numbers are generally similar for the sample of 485 firms in our baseline 2000-2008 panel for China, where 58% of all firms export. On average, Chinese exporters ship 9 HS-8 digit products to 13 markets and, conditional on using foreign inputs, source 33 different products from 6 countries of origin.

Figure 3 illustrates the vast dispersion in average management practices across countries in WMS. The US comes out on top, followed closely by Japan, Germany, Sweden, Canada and the UK. In the middle of the country distribution, Chinese firms are on average significantly less well managed than North American and European companies, but better than firms in Latin America, Africa and other emerging giants such as Brazil and India. These cross-country averages mask substantial variation in management practices across firms in each economy, as shown in Figure 2 for China and the US.

Sample means in Appendix Table 1 corroborate stylized facts in the prior literature that exporters are on average significantly larger and more productive than non-exporters. We document that exporters are on average also better managed than non-exporters: The unconditional export management premium equals 15% of a standard deviation in China and 38% of a standard deviation in the US. In comparison, the export size premia in China and the US stand at 19% and 186% respectively based on firm output and 36% and 123% based on employment.

4 Management Practices and Export Performance

The empirical analysis proceeds in two steps. We first examine the relationship between firms' management practices and export performance. This exercise constitutes a direct test of Propositions 1 and 2. While it informs some of the mechanisms through which management operates, it remains agnostic about the importance of good management for production efficiency and product quality. In Section 5, we study these issues by confronting Propositions 3 and 4 with the data.

We perform the entire analysis separately for China and the US. Given the vast difference in income, institutional quality and factor market frictions between the two countries, this allows us to assess whether management plays a fundamental role in firm activities, and if so, whether its function depends on the specific economic environment.

4.1 Empirical Strategy

To evaluate the empirical validity of Propositions 1 and 2, we investigate the link between firms' management competence and export performance with the following estimating equa-

⁹For the US, we report summary statistics for balance-sheet variables at the establishment level and for trade activity at the firm level, since this is the level at which such data are collected in ASM and LFTTD respectively. The patterns in Appendix Table 1 Panel A look similar if we instead aggregate across establishments up to the firm level.

tion:

$$ExportOutcome_f = \beta Management_f + \Gamma Z_f + \phi_l + \phi_i + \varepsilon_f \tag{11}$$

We consider multiple dimensions of firms' export activity as guided by theory. In different specifications, $ExportOutcome_f$ refers to firm f's exporter status, log global export revenues, and various extensive and intensive margins of exporting. We measure f's managerial competence $Management_f$ with the comprehensive z-score across all management practices surveyed.

We account for any systematic variation in supply and demand conditions across firms in the same location l or industry i with fixed effects, ϕ_l and ϕ_i . These capture differences in factor costs, factor intensities, infrastructure, institutional frictions, tax treatment, etc. that might impact export performance. In the case of China, we add dummies for 31 provinces and 82 sectors based on the primary SIC 3-digit affiliation of each manufacturer. In the case of the US, we use indicator variables for 50 states and about 300 NAICS 6-digit industries.

We further condition on a vector of firm characteristics Z_f . In all specifications, Z_f includes the full set of noise controls pertaining to the management surveys to alleviate potential measurement error in *Management*_f. We subsume the role of firms' ownership type with fixed effects that distinguish between private domestic companies, state-owned enterprises and foreign-owned multinational affiliates in the case of China; firm ownership data is not available for the US. We also report results with an extended set of firm controls Z_f such as firm age, capital and skill intensity, standard productivity measures, and domestic sales. As discussed below, this helps address concerns with omitted variable bias and reverse causality while also shedding light on relevant mechanisms.

The coefficient of interest β reflects the sign of the conditional correlation between firms' management competence and export performance. Given the fixed-effects structure, it is identified from the variation across companies within narrow segments of the economy. This correlation can be interpreted in two ways through the lens of our model. If management corresponds to firms' exogenous productivity draw or one component of it, then β would in principle capture the causal impact of management on export activity. Alternatively, if a primitive firm attribute such as an exogenous productivity draw determines both the choice of management technology and export activity, β would reflect the equilibrium relationship between a production input and output that are joint outcomes of the firm's maximization problem. These two alternatives are isomorphic for our purposes and we do not seek to distinguish between them. Instead, we aim to establish that effective management is a qualitatively and quantitatively important factor in firms' export success (this section), and to examine its role for production efficiency and product quality (Sections 5 and 6).¹⁰

While the US MOPS provides management data for a large cross-section of over 10,000 US

¹⁰Note that reverse causality also does not pose classical estimation bias. If higher export revenues induce firms to adopt better management practices because of economies of scale in management use, this would be consistent with our argument. This mechanism may be amplified if firms learn about novel managerial practices from their experience with foreign buyers and markets (Atkin, Khandelwal, and Osman, 2017).

firms in 2010, WMS covers only about 500 Chinese firms in 2007. In order to fully exploit the information in the Chinese panel data, we therefore estimate specification (11) at the firmyear level, letting all variables but $Management_f$ vary both across firms and over time, and controlling for changes in macroeconomic conditions with year fixed effects ϕ_t . This is motivated by the evidence in Bloom, Mahajan, McKenzie, and Roberts (2018) and patterns in our own MOPS data that management practices evolve slowly within firms over time, such that the crosssectional dispersion dwarfs the time-series variation. We report standard errors clustered by firm since our key variable $Management_f$ is measured at the firm level.

4.2 Export Status, Revenues and Profits

We first establish that better managed firms are significantly more likely to export. Conditional on exporting, they also earn higher export revenues. These findings provide empirical support for Propositions 1 and 2.

Table 1 presents these baseline results. In Columns (1) and (5), we examine firms' export status by setting the dependent variable $ExportOutcome_f$ equal to 1 if a firm lists positive exports on its balance sheets and 0 otherwise. We estimate equation (11) in the matched ASIE-WMS sample for China and the matched ASM-MOPS sample for the US, respectively.¹¹ Firms employing more effective management practices are systematically more likely to enter foreign markets.¹²

We explore the relationship between managerial competence and the scale of export operations in the subset of exporting firms in Columns 3 and 7. We re-estimate specification (11) using the log value of global exports as the outcome variable $ExportOutcome_f$ in the matched CCTS-WMS sample for China and the matched LFTTD-MOPS sample for the US.¹³ We observe that well-run exporters realize substantially higher sales abroad.¹⁴

The strong association between management competence and export activity persists when we add an extended set of firm characteristics Z_f in Columns 2, 4, 6 and 8. We control for firm age using information on the year in which companies were established from ASIE and ASM. We find some evidence that older manufacturers are more likely to be exporters and generate higher export revenues, although these patterns are significant only for the US. We further condition

¹¹For the US, we observe export status at the plant level from ASM and all other trade indicators at the firm level from LFTTD. To exploit the full granularity of the data, we run the baseline US regressions for export status at the plant level. Our results are robust to aggregating up to the firm level, when the coefficient magnitudes increase by 30%-50% in the cross-section in Table 1 and more than double in the panel in Table 7.

¹²We report OLS results, but similar patterns hold with other estimators such as Probit or Logit.

¹³We measure a firm's worldwide exports with the combined value of all its export transactions in the customs records that cover the universe of trade transactions. This arguably gives a more accurate account of exporters' activity than the value of total exports reported on their balance sheets. We have confirmed that the latter produces similar results.

¹⁴Our results for China indicate that multinational companies are more likely to export and have higher export revenues conditional on trading. However, management plays an independent role from foreign ownership that cannot be attributed to multinational affiliates being better managed. State-owned enterprises do not display markedly different outcomes from private domestic firms.

on firms' production technology as reflected in their capital intensity (log net fixed assets per worker) and skill intensity (share of workers with a college degree; log average wage). The results corroborate prior evidence in the literature that more skill- and capital intensive firms are more active exporters, although the point estimates are not always precisely estimated.¹⁵ To guard against omitted variable bias, we always include this broader vector of controls Z_f in the rest of the analysis, but note that the point estimates for $Management_f$ are typically qualitatively and quantitatively close with and without these additional controls.

Our findings point to potentially large economic consequences from improving management practices. Based on our estimates with the extended set of controls, a one-standard-deviation rise in the management z-score is associated with a 5% higher probability of exporting and 23% higher export revenues in China; these numbers are 3% and 37% for the US. Given the large management gaps across countries shown in Figure 3, this implies that variations in management competence could account for substantial differences in trade intensity across countries. These magnitudes are also sizeable compared to the role of firm age, skill- and capital intensity (comparable statistics for these are in the range of 2% to 28%).

In addition to export status and revenues, Proposition 2 also has implications for firms' export profits. As standard with balance-sheet data, however, we observe firms' consolidated profits from worldwide sales that cannot be broken down by market. In Appendix Table 2, we exploit the available information as best we can, and find indicative evidence of a positive link between effective management and export profits. We first confirm that superior managerial practices are associated with higher firm profits, with and without the expanded set of firm controls (Columns 1-2 and 4-5). We then document that this holds even conditioning on domestic sales, calculated as the difference between total turnover and total exports (columns 3 and 6).

4.3 Extensive and Intensive Export Margins

As a first step to understanding the mechanisms through which management contributes to export success, we decompose exporters' trade activity into the number of foreign markets they enter and the sales they make in each market. We find that better managed firms have the capacity both to serve more export markets and to sell more in individual markets.

We measure the extensive margin of firms' exports with the log number of destination countries they supply, the log number of products they ship to at least one destination, and the log total number of destination-product markets they penetrate. We quantify the intensive margin with average log exports per destination-product. We define products at a very granular level, namely HS 8-digit categories. We re-estimate equation (11) using each export margin in place of *ExportOutcome*_f, and report our findings in Table 2. Appendix Table 3 contains symmetric regressions without the wider set of firm controls Z_f .

¹⁵The positive correlation between average wages and the share of skilled workers across Chinese firms generates multicollinearity in Columns 2 and 4 and a negative coefficient on the skilled labor share. Both measures of skill intensity enter positively and significantly when included one at a time.

We consistently observe positive coefficients on $Management_f$ across all specifications that are statistically significant in all but one case (the intensive margin in China). For Chinese firms, a one-standard-deviation improvement in managerial competence is associated with 19% more export destinations, 17% more export products, 22% more destination-product markets, and 2% higher exports in the average market (Columns 1-4 of Table 2). For American companies, these magnitudes stand respectively at 13%, 17%, 20%, and 18% (Columns 6-9).

Overall, the extensive margin of market entry accounts for just over half of the contribution of management to firm exports in the US. In the case of China, this share reaches 90% when we condition on the full set of firm controls and 75% when we do not.¹⁶

These results are in line with the theoretical predictions for the margins of firms' export activity summarized in Proposition 2. As a final check on internal consistency, we consider the variation in export sales across a firm's destination-product markets. In our model, exporters add foreign markets in decreasing order of profitability. As a result, better managed firms servicing more markets do so by entering progressively smaller markets where they earn lower sales. This composition effect implies that our intensive-margin results underestimate the relationship between management and exports to any given market. Further analysis supports this. For each firm, we identify its largest destination-product market by sales revenues, and regress log exports to this top market on $Management_f$. We obtain much larger coefficients than those for the intensive margin that are moreover significant for both China and the US (Columns 5 and 10). As we repeat this exercise replacing the outcome variable with log average sales to the top two, top three, etc. export markets, we record progressively lower point estimates as anticipated.

4.4 Exports vs. Domestic Activity

We are interested in whether the positive association between management quality and export performance reflects a general beneficial effect of good management on firm activity. Through the lens of our model, effective management practices improve firm performance both at home and abroad, such that better managed firms have higher domestic sales, higher probability of exporting, and higher export revenues. The elasticities of these three outcomes with respect to management differ and, as with productivity elasticities in workhorse trade models, generally depend on modeling assumptions about demand.¹⁷ In our CES set-up, better management increases firm revenues proportionately in all markets served, but it also induces entry into more markets. As a result, total exports rise faster with management competence than domestic sales.

Appendix Table 4 corroborates these patterns in the data, further validating our model. We compute firms' log domestic sales by taking the difference between total sales and total exports as reported on companies' balance sheets and matched customs records. Columns 1 and 6 confirm

¹⁶These calculations are based on comparing regression coefficients across specifications for different export outcomes, such as Column 8 of Table 1 and Column 8 of Table 2.

¹⁷For example, the ratio of a firm's sales in two markets is independent of firm productivity with CES but not with linear demand or with non-homothetic preferences (Melitz, 2003; Melitz and Ottaviano, 2008).

that producers with advanced management practices sell more at home. In the rest of Appendix Table 4, we repeat the main regressions for manufacturers' export status, global export revenues and various export margins controlling for their domestic sales in addition to the extended set of firm characteristics Z_f . We continue to record positive significant coefficients on $Management_f$ (except for average exports per destination-product for China as before).

4.5 Interpretation: Management as Productivity

The results indicate that successful export performance is closely related to the use of sophisticated management practices. Through the lens of the model, we interpret this as evidence that managing capital and labor resources effectively is critical to firm productivity. In this sub-section, we explore this management-productivity nexus.

The theoretical notion of productivity in the literature is quantity-based total factor productivity TFPQ. In our model, this corresponds to firms' capacity to produce a given quantity and quality of output at lower cost. We thus view management competence as a measurable, tangible counterpart to the theoretical concept of TFPQ, or at least an important component of TFPQ.

Standard revenue-based proxies for productivity, TFPR, are constructed from data on sales revenues and input costs (capital, labor, materials). This approach faces two challenges. First, TFPR is a noisy measure of TFPQ because it incorporates input and output prices and markups (Hsieh and Klenow, 2009; De Loecker, 2011; Bartelsman, Haltiwanger, and Scarpetta, 2013). This introduces bias in regressions of firm outcomes such as export activity on TFPR. Second, TFPR constitutes a residual from a production function and is thus a black box with no precise economic content or actionable policy implication.

Being able to observe and quantify management practices helps overcome both of these challenges. Management effectiveness identifies specific practices that firms use in production, such as setting targets, monitoring operations, and incentivizing workers. This unpacks the black box of TFPR residuals to isolate well-defined economic mechanisms. Management measures also circumvent estimation biases associated with TFPR since they are obtained independently from data on firms' production and trade activity.

Question 1: Where does good management come from? Observed management practices may have both an exogenous and an endogenous component. On the one hand, management competence may be an exogenous draw at the firm level as in our baseline model. In this case, β in specification (11) would capture the causal impact of management quality on firm performance from, for example, exogenous variation in managers' ability or style (Bertrand and Schoar, 2003).

On the other hand, firms may endogenously choose their managerial quality based on an exogenous firm primitive. Estimates of β would then reflect the equilibrium relationship between joint outcomes of the firm's maximization problem, for example if exogenously different founderentrepreneurs endogenously hire managers of different skill levels. While we do not distinguish between these two theoretical alternatives, we establish that management matters for firm performance and uncover the mechanisms through which it operates.

Question 2: How does management relate to TFP? Heterogeneous-firm theory traditionally focuses on TFPQ productivity as the firm attribute that uniquely determines all outcomes of the firm's problem, including all aspects of trade activity. In practice, evidence for many countries indicates that measured TFPR is positively but imperfectly correlated with measured firm outcomes. There are two possible explanations for this discrepancy: TFPR and/or firm outcomes are measured with error, or multiple firm attributes matter. For example, two different draws may fix firms' cost of producing respectively physical units and quality, and aggregate into composite capability that pins down firm outcomes (Hallak and Sivadasan, 2013). We investigate the relationship between management practices and firm productivity in Appendix Table 5. We construct $TFPR_f$ as in Levinsohn and Petrin (2003) using balance-sheet data on companies' total sales, capital expenditures, labor costs and material purchases, and accounting for differences in production technology across industries and ownership types.

We first estimate the conditional correlation between $Management_f$ and $TFPR_f$ by setting $TFPR_f$ as the left-hand side variable in specification (11) (Column 1). TFPR is indeed higher in firms employing more sophisticated management practices. We then replicate regression (11) for $TFPR_f$ in place of $Management_f$ (Columns 2-3). TFPR systematically enters positively and significantly, except for Chinese firms' export status.

Finally, we decompose TFPR into two components by regressing $TFPR_f$ on $Management_f$ with no other controls: the projection onto $Management_f$ and the residual term which we call $nonManagementTFPR_f$. In Columns 4-5, we compare the contribution of $Management_f$ and $nonManagementTFPR_f$ to the variation in trade outcomes by adding both to the right-hand side of equation (11). The estimates for $Management_f$ are very similar to the baseline in Table 1, and imply that the management component of TFP has economically large explanatory power in absolute and in relative terms. The rest of Appendix Table 5 documents similar patterns for indicators of product quality and production efficiency which we introduce in Section 5.

Question 3: Which management components matter most? A policy-relevant question is whether certain management practices are more instrumental to firm performance than others. While the baseline management z-score we use is an average across all practices surveyed, we can unbundle this average into sub-components in order to shed light on this policy question. To this end, we distinguish between "targets & monitoring" (information collection and processing) and "incentives" (hiring, firing, pay and promotions). The results in Appendix Table 7 indicate that both sets of practices are significant, although perhaps with slightly stronger effects on the targets & monitoring practices.

5 Management Mechanisms: Efficiency and Quality

5.1 Structural Estimates

Having established that advanced managerial practices are associated with superior export performance, we next assess the empirical validity of Propositions 3 and 4 to inform the underlying mechanisms through which management operates. In particular, we are interested in whether effective management improves firms' production efficiency, capacity to manufacture high-quality products, or both. The results we establish lead us to conclude that management acts through both these efficiency and quality channels.

We first consider the predictions of Proposition 3 for the relationship between firms' management practices, product quality, and quality-adjusted prices. We exploit the rich dimensionality of the data and examine firms' behavior in finely disaggregated export markets. This allows us to study the role of management while accounting for various supply and demand conditions with an extensive set of fixed effects in the following estimating equations:

$$\ln(Quality_{fdp}) = \beta^q Management_f + \Gamma^q Z_f + \phi_l^q + \phi_{dp}^q + \varepsilon_{fdp}^q$$
(12)

$$\ln(Price_{fdp}/Quality_{fdp}) = \beta^{p/q}Management_f + \Gamma^{p/q}Z_f + \phi_l^{p/q} + \phi_{dp}^{p/q} + \varepsilon_{fdp}^{p/q}$$
(13)

Through the lens of our model, the coefficient on management in the quality equation (β^q) identifies the structural parameter θ , which governs the effect of management on product quality. Similarly, the coefficient on management in the quality-adjusted price equation $(\beta^{p/q})$ identifies the structural parameter δ , which captures the effect of management on productive efficiency. Specifications (12) and (13) reveal whether management operates through both the quality and the efficiency mechanisms: According to Proposition 3, we should observe $\beta^q > 0$ and $\beta^{p/q} < 0$ if and only if the quality and efficiency channels are active, respectively. Moreover, this interpretation is conservative given the potential for variable mark-ups.¹⁸

The unit of observation is now the firm-destination-HS8 product(-year).¹⁹ Price_{fdp} is the export unit value that firm f charges for product p in destination country d (in year t). We use free-on-board export prices that exclude trade duties, transportation costs and retailers' mark-up, such that $Price_{fdp}$ corresponds to the sum of the exporter's marginal cost and mark-up. We construct model-consistent proxies for firms' export product quality and quality-adjusted price using data on export prices and quantities by firm, product, destination (and year). As discussed in Section 2.1, $\ln q_{ji} \propto \sigma \ln p_{ji}^{fob} + \ln x_{ji}$, such that \log quality $\ln q_{ji}$ can be inferred as the sum of log quantity x_{ji} and log free-on-board price p_{ji}^{fob} , adjusted for the elasticity of substitution across

¹⁸If better managed firms set higher mark-ups, our conclusions for β^q would be unaffected, but p_{ji}^{fob}/q_{ji} would be inflated and we would be less likely to find $\beta^{p/q} < 0$.

¹⁹All results for China hold when we distinguigh between processing and ordinary exports. We find similar patterns when we consider the firm–destination–product-trade regime–year quintuplet as the unit of observation and include a complete set of destination–product-trade regime triple fixed effects.

varieties σ . We set $\sigma = 5$ (the median value in calibration exercises in the prior literature), but our results are robust to alternative assumptions about this elasticity (Khandelwal, Schott, and Wei, 2013).

We continue to include fixed effects for firms' provincial or state location ϕ_l and the full set of firm controls Z_f , as well as year fixed effects for China. Instead of the fixed effects for firms' primary industry ϕ_i in equation (11), we now condition on destination-product pair fixed effects ϕ_{dp} . These subsume the variation in total expenditure, consumer price indices and trade costs across countries and products in the model, as well as any observable and unobservable differences in consumer preferences, institutional frictions and other forces outside the model. Specifications (12) and (13) are thus a very stringent test of our theory, as the coefficient on $Management_f$ is identified from the variation across firms within very narrow segments of the global economy, such as Chinese exporters of men's leather shoes to Germany or US exporters of cellular phones to Japan. We conservatively cluster standard errors by firm to accommodate correlated shocks across destinations and products within firms.

Equations (12) and (13) are in the spirit of prior studies of the relationship between measured firm productivity (TFPR), prices and revenues (Kugler and Verhoogen, 2009; Manova and Zhang, 2012). Since these variables are all constructed from the same raw data on sales and quantities, a common challenge in this literature has been ruling out estimation biases arising from correlated non-classical measurement error in the right- and left-hand side variables. We circumvent this problem by using direct measures of management practices that are entirely independent of the sales and quantity data.

The evidence in Table 3 lends strong support to managerial competence improving both production efficiency and product quality. In both China and the US, we observe that management is associated with significantly higher export quality (Columns 1 and 5) and significantly lower quality-adjusted prices (Columns 2 and 6). Formally, we find that $\theta^{CH} = 0.531$, $\delta^{CH} = 0.385$, $\theta^{US} = 0.048$ and $\delta^{US} = 0.045$. Based on these estimates, upgrading management practices by one standard deviation is associated with a 53% increase in product quality and a 39% decline in quality-adjusted prices in China. In the case of the US, quality and quality-adjusted prices are equally elastic with respect to management competence: a one-standard-deviation rise in the management score is accompanied with roughly a 5% change in both.

Comparing the magnitude of the management effects across countries should be done with caution, because of the different survey methodologies and sample sizes. However, the relative effect of management on quality and efficiency within each country is arguably informative. We draw two conclusions from our results.

First, management appears to have a larger impact on both productive efficiency and product quality in China than in the US, $\delta^{CH} > \delta^{US}$ and $\theta^{CH} > \theta^{US}$. One possible explanation is diminishing returns to management, since management practices are on average substantially worse in China: Initial improvements in management yield large gains, but additional improvements address more and more marginal issues with progressively smaller incremental benefit to quality

and efficiency.

Second, the parameters suggest that management has a relatively bigger effect on product quality than on productive efficiency in China compared to the US, $\theta^{CH} - \delta^{CH} > \theta^{US} - \delta^{US} = 0$. We explore this further by directly estimating the following price equation in Columns 3 and 7:

$$\ln(Price_{fdp}) = \beta^p Management_f + \Gamma^p Z_f + \phi_l^p + \phi_{dp}^p + \varepsilon_{fdp}^p$$
(14)

Our theory implies that $\beta^p = \beta^q - \beta^{p/q} = \theta - \delta$ reflects the relative impact of management on quality vs. efficiency. Consistently with the findings from specifications (12) and (13), the relationship between prices and management is significantly positive in China and insignificantly different from 0 in the US. The greater impact of management on quality relative to efficiency in China is intuitive. When quality levels are relatively low (as they are in China compared to the US), a marginal change in managerial competence is likely to have much larger impacts on product quality. This is consistent with the hypothesis of Sutton (2007) that moving up the product quality ladder through improved management practices is critical for emerging economies.

For completeness, Columns 4 and 8 document the elasticity of export quantity with respect to management. In the model, this elasticity is $\delta\sigma - \theta$ and its sign theoretically ambiguous. In practice, quantity is invariant with management effectiveness in China and increasing in the US.

5.2 Robustness

In this subsection, we perform several robustness checks and extensions to alleviate concerns with alternative interpretations of the results for export prices and product quality.

5.2.1 Demand Elasticity

In order to infer a model-consistent proxy for product quality $\ln q_{ji} \propto \sigma \ln p_{ji}^{fob} + \ln x_{ji}$, we need the price elasticity of demand σ , which with CES preferences corresponds to the elasticity of substitution across varieties in consumption. While our baseline analysis takes a standard parameter value from the literature, $\sigma = 5$, our findings are robust to alternative assumptions. We have confirmed that qualitatively similar patterns obtain when we instead set σ equal to 4, 7, or 10. The results also remain unchanged when we allow σ to vary across SIC 3-digit industries using estimates from Broda and Weinstein (2006) (Panel A of Appendix Table 6).

5.2.2 Variable Mark-ups

Management practices may affect not only production efficiency and product quality, but also firms' mark-ups and thereby prices. This channel is moot in our model because CES preferences generate constant mark-ups, but it may be important in practice. Consider a world with no quality differentiation across firms. The prior theoretical literature has shown that in certain environments with variable mark-ups, more productive firms charge lower prices even though they set higher mark-ups (Melitz and Ottaviano, 2008; Eaton and Kortum, 2002). With alternative market structures or strategic behavior, however, mark-ups could in principle rise sufficiently quickly with firm productivity to dominate the associated decline in marginal costs and result in higher prices.

In the presence of quality differentiation across firms, variable mark-ups might therefore confound the inference of product quality from price and quantity data. In the former case, they might lead us to underestimate the impact of good management on production efficiency, while overestimating its effect on product quality. In the latter case, our findings for inferred quality and quality-adjusted prices might be driven by better managed firms extracting higher mark-ups rather than offering more sophisticated products that they assemble more efficiently.

As a step towards addressing this concern, we establish that similar results hold when we control for firms' market share as a proxy for their ability to impose higher mark-ups (Panel B of Appendix Table 6). We use a Chinese (US) firm's share of total Chinese (US) exports to a given destination-product, $\frac{Exports_{fdp}}{\sum_{f} Exports_{fdp}}$, as an indicator of its market power in that market. The results for management remain robust.

5.3 Input Characteristics

We next test the predictions of Proposition 4 for the quality of firms' intermediate inputs and the complexity of their assembly technology. Since we do not directly observe input quality and assembly complexity in the data, we proxy them with a variety of observed input characteristics. We construct these using balance-sheet data on firms' total material purchases and customs records on the universe of firms' imported input purchases by product and country of origin; as common with production data, we cannot access information on firms's domestic inputs.

We estimate specifications of the following two types:

$$InputCharacteristic_f = \beta Management_f + \Gamma Z_f + \phi_l + \phi_i + \varepsilon_f \tag{15}$$

$$InputCharacteristic_{fop} = \beta Management_f + \Gamma Z_f + \phi_l + \phi_{op} + \varepsilon_{fop}$$
(16)

As in equation (11), the unit of observation in regression (15) is the firm, and we include the same set of controls (location and industry fixed effects; full set of noise and firm controls). Similar to equation (12), the unit of observation in regression (16) is the firm-country of origin-product, and we include the same set of controls (location fixed effects; country of origin-product pair fixed effects; full set of noise and firm controls). In the case of China, we again exploit the panel and add year fixed effects. We cluster error terms by firm as before.

5.3.1 Input Quality

The quality production technology in the model stipulates that making goods of higher quality is associated with higher marginal costs. One possibility is that this reflects the need for highquality intermediate inputs.²⁰ Table 4 provides evidence consistent with better managed firms sourcing more expensive, higher-quality inputs from richer countries of origin $(\theta > 0)$.²¹ In Columns 1-2 and 6-7, we estimate regression (15) for the log value of imports and the log share of imports in total input purchases. For both China and the US, we find that better managed firms have higher imports, consistent with their operating on a bigger scale and using more inputs in absolute terms. Unlike American producers, however, better managed Chinese producers also import a systematically higher share of their inputs, in line with priors about the paucity of specialized, high-quality domestic inputs in China. The insignificant estimates for the US thus serve as a corroborating placebo test.

Columns 3 and 8 confirm that well-run companies in both China and the US buy inputs from suppliers located in richer, more developed economies. Such economies are believed to produce higher-quality, more sophisticated goods because they employ advanced technologies and more skilled workers (Schott, 2004). In these specifications, the outcome variable is the weighted average log GDP per capita across a firm's foreign suppliers, using imports by origin country as weights. A one-standard-deviation rise in management competence is associated with 4%-5% higher average source-country income.

In Columns 4 and 9 of Table 4, we estimate regression (16) for the log unit value of firm imports by product and country of origin. Advanced management practices are accompanied by higher imported input prices in China, but not significantly so in the US. In Columns 5 and 10, we apply the structural transformation to import unit values to obtain inferred input quality, in the same manner as we did with inferred export quality. We find that better managed firms in both countries use higher-quality imported inputs, with a significantly higher elasticity for China than for the US. Improving management effectiveness by one standard deviation corresponds respectively to 10% and 58% higher imported input price and quality among Chinese manufacturers, but only 0% and 5% among their US counterparts. These results suggest that at lower levels of management competence and product quality - such as the Chinese context - good management can help firms to not only more effectively source and process inputs from advanced countries, but also to better identify high-quality suppliers within each origin country. This additional channel might contribute towards the significantly higher elasticity of output quality with respect to management that we documented above for China relative to the US.

5.3.2 Assembly Complexity

A body of work has proposed that manufacturing more sophisticated products entails the assembly of a wider range of specialized inputs, possibly through the completion of more manufacturing

²⁰Recall the dress example: A garment producer can choose what materials to use in order to make a dress according to preset designs and assembly steps. He could use cheap cotton and plastic buttons to make a cheap, low-quality dress or expensive silk and mother-of-pearl buttons to make an expensive, high-quality dress.

 $^{^{21}}$ As we show in Appendix 2.3, one justification for the quality production function in our model is complementarity between input quality and management competence in the production of output quality. We find some evidence consistent with this mechanism in unreported results for the US.

stages (Hummels, Ishii, and Yi, 2001; Yi, 2003; Johnson and Noguera, 2012). This provides a second possible rationalization for the quality production function in the model. We therefore use the variety of a firms' imported inputs as a proxy for the complexity of their assembly technology. We also account for product differentiation across countries supplying the same product (Krugman, 1980). In particular, we measure input variety with the log number of different HS-8 products, countries of origin, or origin country-product pairs in a firm's import portfolio, and estimate specification (15) for each of these measures. As Table 5 demonstrates, better managed companies systematically source more distinct inputs from more suppliers, in terms of more origin-product combinations. These results obtain conditioning on firms' log number of export products. This ensures that the range of imported inputs does not rise with management competence simply because of a commensurate increase in the number of output products, rather than the use of more complex assembly.

In light of Proposition 4, the patterns in Tables 4 and 5 corroborate the idea that effective management enables firms to produce higher-quality products using higher-quality inputs and more complex production processes. Intuitively, this could be attributed to good management improving quality control and reducing the incidence of costly mistakes in manufacturing, which is especially relevant when using expensive, high-quality inputs. Superior management may also enhance the processing of specialized inputs that need to be mutually compatible for final assembly, the coordination of multiple production stages, and the implementation of efficient inventory practices. These practices are particularly important when the manufacturing process is more complex.

6 Causal Effect of Management

The model in Section 2 has a range of empirical implications for the co-variation we expect to see between firms' management practices and multiple aspects of their trade activity. These are broadly supported by the data. Although we can extend the model to capture endogenous choice of management practice (see Appendix 2.1), we cannot empirically test this alternative model as its predictions are observationally equivalent to those of the baseline framework with exogenous management draws.

In this section, we provide two pieces of independent evidence which suggest that management competence indeed exerts a causal effect on firms' production efficiency and quality capacity, and thereby on their export performance: one based on a management intervention in a randomized control trial in India, and one based on the panel dimension of our data for the US.

6.1 RCT Evidence

A growing body of work using randomized control trials (RCT) indicates that some aspects of management have a causal effect on firm TFP, production efficiency and product quality (e.g. see the survey in McKenzie and Woodruff (2013)). Perhaps the best evidence comes from Bloom,

Eifert, Mahajan, McKenzie, and Roberts (2013), who worked with the company Accenture to provide free management consulting services to large firms (average of 273 employees) in the textile industry in Mumbai, India. The study examined three groups of plants over the 2008-2011 period. 11 plants owned by 6 firms served as a pure control group, while 20 plants owned by 11 firms constituted the treatment group. In the treated group, 14 plants were randomly selected to receive the management intervention. They had one month of diagnostic assessment of management practices in place and four months of consulting on 38 core management practices across 6 key areas (factory operations, quality control, inventory control, loom planning, human resources, and sales and orders). The remaining 6 plants in the treated firms were given only the one-month diagnostic without any intervention. All three groups of plants were followed for a further 3 years with monthly visits to collect detailed production data. In 2017, Bloom, Mahajan, McKenzie, and Roberts (2018) went back to these firms to assess the long-term impact of the intervention 8 years on. They collected various follow-up performance metrics for 2014 and 2017, including trade activity that we are the first to analyze here.

Three lessons emerge from the India RCT. First, the consulting intervention had a large long-lasting effect on the management practices that firms actually adopted. The management practice adoption rate in the treatment plants rose from 25.6% to 63.4% in the first year, with this slipping back somewhat over the next eight years back to 46%, but still significantly above its initial level (or the control firms).

Second, the management intervention led to a large causal improvement in firms' TFP and product quality. Figure 4 plots the change in TFP during the experiment against the change in management competence for both treatment and control plants, displaying a highly significant positive relationship. Figure 5 similarly plots the change in the product defect rate (an inverse measure of quality) against the change in the management score, and shows a strongly significant downward slope. As Bloom, Eifert, Mahajan, McKenzie, and Roberts (2013) discuss, the intervention led to a 37.8% improvement in management effectiveness. This caused a 43% drop in product defects, and was itself one of the major drivers of the 17% increase in TFP.

Third, the management intervention significantly increased firms' export participation. In the 2017 long-run follow-up, we asked firms about their export activity over the prior 10 years. In Panel A of Table 6, we explore the intention-to-treat effect of the management consulting in a reduced-form regression of various export outcomes on a plant-level treatment dummy. We see in Column 1 that treatment plants were 0.189 more likely to export in the post-treatment period, suggesting that better management practices increase the extensive margin of exporting. In Columns 2 and 3, we examine export levels respectively with the log of (1+exports) or (exports), and find highly significant positive effects for both. The results for the pure intensive margin in Column 3 imply a 51.6% increase in export revenues from the management intervention. We document similarly strong positive impacts in Panel B, where we use the treatment indicator as an instrument for the management score in a two-stage IV specification.

These three pieces of causal RCT evidence highlight how adopting superior management

strategies can lead to improvements in firms' product quality, production efficiency, and ultimately export performance in environments with weak management practices such as India.

6.2 Panel Evidence

We shed further light on the possible causal effect of firm management on trade activity by exploiting the panel dimension in our data for the US. In Panel A of Table 7, we replicate all of our key results when regressing firms' export and import outcomes in year 2011 on their lagged management score (from year 2010). Conditioning as before on the full set of state fixed effects, industry fixed effects, noise and firm controls, we establish broadly similar results.

Using even more stringent estimation, we show that within-firm upgrading of management practices is associated with significant improvements in export performance, production efficiency and output quality, and imported input quality and complexity. In particular, in Panel B of Table 7 we regress the change in all relevant trade outcomes within firms from 2005 to 2010 on the concurrent change in their management practices.²² We include state and industry fixed effects, which now absorb not only level differences, but also systematic differences in time trends across space and manufacturing sectors. Controlling for the full set of firm and noise controls as above, we record significant coefficients across the board. Their magnitudes are occasionally slightly reduced, consistent with management exerting greater effects on the level of firm performance than on its growth trajectory. The only point estimate that changes sign in these first-difference regressions is that for the average GDP per capita of firms' source countries. We believe this may be related to the rise of China over the 2005-2010 period - a country with relatively low income but nevertheless steadily increasing product quality over time.

7 Conclusion

This paper examines for the first time the role of management practices for firms' export performance. We theoretically and empirically establish that management ability can affect both productive efficiency and quality capacity. Good managerial practices thereby enhance export participation, intensity and number of markets served. These results suggest that effective management is an important aspect of firm productivity which has typically been treated as a black box in the prior literature. We show that although better management is associated with greater efficiency and quality capacity in both China and the US, it matters relatively more in China, and especially so for the quality channel. This is consistent with the idea that enhanced managerial capabilities are critical for helping emerging economies move up the quality ladder and become richer (Sutton, 2007).

 $^{^{22}}$ In 2010, US firms completing MOPS were asked about their management practices in both 2005 and 2010. In 2015, they likewise reported on their management practices in 2015 and 2010. The contemporaneous and recall data for 2010 line up well (Bloom, Brynjolfsson, Foster, Jarmin, Patnaik, Saporta-Eksten, and Van Reenen, 2017), which gives us confidence in using recall data.

Our findings have broader implications for the microeconomics of firm operations and inform active literatures on the nature, origin and welfare consequences of firm heterogeneity. They also speak to policy concerns about the impact of limited management know-how on growth and entrepreneurship in developing economies. A limitation of the work is that we have not examined the reasons for weaker managerial ability in some firms and countries compared to others. We believe that strengthening managerial capabilities is an important policy issue, and researchers are starting to make in-roads into the key question of what determines management and how policy-makers can influence it.

References

- Ackerberg, D. A., K. Caves, and G. Frazer (2015). Identification properties of recent production function estimators. *Econometrica* 83(6), 2411–2451.
- Arkolakis, C., A. Costinot, and A. Rodríguez-Clare (2012). New trade models, same old gains? The American Economic Review 102(1), 94–130.
- Atkin, D., A. K. Khandelwal, and A. Osman (2017). Exporting and firm performance: evidence from a randomized experiment. The Quarterly Journal of Economics 132(2), 551–615.
- Baldwin, R. and J. Harrigan (2011). Zeros, quality, and space: Trade theory and trade evidence. American Economic Journal: Microeconomics 3(2), 60–88.
- Bartelsman, E., J. Haltiwanger, and S. Scarpetta (2013). Cross-country differences in productivity: The role of allocation and selection. *The American Economic Review* 103(1), 305–334.
- Bastos, P., J. Silva, and E. Verhoogen (2018). Export destinations and input prices. American Economic Review 108(2), 353–92.
- Bernard, A. B., J. Eaton, J. B. Jensen, and S. Kortum (2003). Plants and productivity in international trade. *The American Economic Review* 93(4), 1268–1290.
- Bernard, A. B., J. B. Jensen, S. J. Redding, and P. K. Schott (2007). Firms in international trade. *Journal of Economic Perspectives* 21(3), 105–130.
- Bernard, A. B., S. J. Redding, and P. K. Schott (2010). Multiple-product firms and product switching. *The American Economic Review* 100(1), 70–97.
- Berthou, A., J. J. Chung, K. Manova, and C. S. D. Bragard (2016). Productivity, misallocation and trade. *University of Oxford Mimeo*.
- Bertrand, M. and A. Schoar (2003). Managing with style: The effect of managers on firm policies. The Quarterly Journal of Economics 118(4), 1169–1208.
- Bloom, N., E. Brynjolfsson, L. Foster, R. S. Jarmin, M. Patnaik, I. Saporta-Eksten, and J. Van Reenen (2017). What drives differences in management? Technical report, National Bureau of Economic Research.
- Bloom, N., B. Eifert, A. Mahajan, D. McKenzie, and J. Roberts (2013). Does management matter? evidence from india. *The Quarterly Journal of Economics* 128(1), 1–51.
- Bloom, N., A. Mahajan, D. McKenzie, and J. Roberts (2018). Do management interventions last? evidence from india. Technical report, National Bureau of Economic Research.

- Bloom, N. and J. Van Reenen (2007). Measuring and explaining management practices across firms and countries. *The Quarterly Journal of Economics* 122(4), 1351–1408.
- Broda, C. and D. E. Weinstein (2006). Globalization and the gains from variety. *The Quarterly Journal of Economics* 121(2), 541–585.
- Bustos, P. (2011). Trade liberalization, exports, and technology upgrading: Evidence on the impact of mercosur on argentinian firms. *The American Economic Review* 101(1), 304–340.
- Caliendo, L., F. Monte, and E. Rossi-Hansberg (2017). Exporting and organizational change. Technical report, National Bureau of Economic Research.
- Caliendo, L. and E. Rossi-Hansberg (2012). The impact of trade on organization and productivity. The quarterly journal of economics 127(3), 1393–1467.
- Chakraborty, P. and O. Raveh (2018). Input-trade liberalization and the demand for managers: Evidence from india.
- Chen, C. and C. Steinwender (2016). Import competition, heterogeneous preferences of managers and productivity. Technical report, mimeo, December 2016.[14].
- De Loecker, J. (2011). Product differentiation, multiproduct firms, and estimating the impact of trade liberalization on productivity. *Econometrica* 79(5), 1407–1451.
- Deming, W. E. (1950). Elementary principles of the statistical control of quality: a series of lectures. Nippon Kagaku Gijutsu Remmei.
- Drew, J., B. McCallum, and S. Roggenhofer (2016). *Journey to Lean: Making Operational Change Stick*. Springer.
- Eaton, J. and S. Kortum (2002). Technology, geography, and trade. *Econometrica* 70(5), 1741–1779.
- Eckel, C., L. Iacovone, B. Javorcik, and J. P. Neary (2015). Multi-product firms at home and away: Cost-versus quality-based competence. *Journal of International Economics* 95(2), 216– 232.
- Eckel, C. and J. P. Neary (2010). Multi-product firms and flexible manufacturing in the global economy. The Review of Economic Studies $\gamma\gamma(1)$, 188–217.
- Fieler, A. C., M. Eslava, and D. Y. Xu (2018). Trade, quality upgrading, and input linkages: Theory and evidence from colombia. *American Economic Review* 108(1), 109–46.
- Goldberg, P. K., A. K. Khandelwal, N. Pavcnik, and P. Topalova (2010). Imported intermediate inputs and domestic product growth: Evidence from india. *The Quarterly Journal of Economics* 125(4), 1727–1767.

- Group, R. (2014). The Lean Healthcare Dictionary: An Illustrated Guide to Using the Language of Lean Management in Healthcare. CRC Press.
- Hallak, J. C. and J. Sivadasan (2013). Product and process productivity: Implications for quality choice and conditional exporter premia. *Journal of International Economics* 91(1), 53–67.
- Hsieh, C.-T. and P. J. Klenow (2009). Misallocation and manufacturing tfp in china and india. The Quarterly Journal of Economics 124(4), 1403–1448.
- Hummels, D., J. Ishii, and K.-M. Yi (2001). The nature and growth of vertical specialization in world trade. *Journal of international Economics* 54(1), 75–96.
- Johnson, R. C. (2012). Trade and prices with heterogeneous firms. Journal of International Economics 86(1), 43–56.
- Johnson, R. C. and G. Noguera (2012). Accounting for intermediates: Production sharing and trade in value added. *Journal of International Economics* 86(2), 224–236.
- Khandelwal, A. (2010). The long and short (of) quality ladders. The Review of Economic Studies 77(4), 1450–1476.
- Khandelwal, A. K., P. K. Schott, and S.-J. Wei (2013). Trade liberalization and embedded institutional reform: Evidence from chinese exporters. *The American Economic Review* 103(6), 2169–2195.
- Krugman, P. (1980). Scale economies, product differentiation, and the pattern of trade. The American Economic Review 70(5), 950–959.
- Kugler, M. and E. Verhoogen (2009). Plants and imported inputs: New facts and an interpretation. The American Economic Review 99(2), 501–507.
- Kugler, M. and E. Verhoogen (2012). Prices, plant size, and product quality. The Review of Economic Studies 79(1), 307–339.
- Levinsohn, J. and A. Petrin (2003). Estimating production functions using inputs to control for unobservables. The Review of Economic Studies 70(2), 317-341.
- Manova, K. and Z. Yu (2017). Multi-product firms and product quality. Journal of International Economics 109, 116–137.
- Manova, K. and Z. Zhang (2012). Export prices across firms and destinations. The Quarterly Journal of Economics, 379–436.
- McKenzie, D. and C. Woodruff (2013). What are we learning from business training and entrepreneurship evaluations around the developing world? The World Bank Research Observer 29(1), 48–82.

- Melitz, M. J. (2003). The impact of trade on intra-industry reallocations and aggregate industry productivity. *Econometrica* 71(6), 1695–1725.
- Melitz, M. J. and G. I. Ottaviano (2008). Market size, trade, and productivity. The Review of Economic Studies 75(1), 295–316.
- Melitz, M. J. and S. J. Redding (2013). Firm heterogeneity and aggregate welfare.
- Myerson, P. (2014). Lean Retail and Wholesale. McGraw-Hill Education.
- Pavcnik, N. (2002). Trade liberalization, exit, and productivity improvements: Evidence from chilean plants. *The Review of Economic Studies* 69(1), 245–276.
- Schott, P. K. (2004). Across-product versus within-product specialization in international trade. The Quarterly Journal of Economics 119(2), 647–678.
- Sutton, J. (2007). Quality, trade and the moving window: The globalisation process. The Economic Journal 117(524), F469–F498.
- Sutton, J. (2012). Competing in Capabilities: the Globalization Process. Oxford University Press, Oxford.
- Syverson, C. (2011). What determines productivity. Journal of Economic literature 49(2), 326-365.
- Taylor, F. W. (1911). The Principles of Scientific Management. Harper.
- Teeuwen, B. (2010). Lean for the Public Sector: The Pursuit of Perfection in Government Services. CRC Press.
- Verhoogen, E. A. (2008). Trade, quality upgrading, and wage inequality in the mexican manufacturing sector. The Quarterly Journal of Economics 123(2), 489–530.
- Walker, F. A. (1887). The source of business profits. *The Quarterly Journal of Economics* 1(3), 265–288.
- Womack, J. P., D. T. Jones, and D. Roos (1990). *Machine that Changed the World*. Simon and Schuster.
- Yi, K.-M. (2003). Can vertical specialization explain the growth of world trade? Journal of Political Economy 111(1), 52–102.

		Chi	na			U	S	
Dep Variable:	Exporte	r Dummy	Log E	Exports	Exporter	Dummy	Log E	xports
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Management	0.040** (2.30)	0.048*** (2.75)	0.260** (2.14)	0.231* (1.81)	0.042*** (13.92)	0.031*** (10.13)	0.488*** (21.72)	0.373*** (16.79)
Capital Intensity		-0.010 (-0.76)		0.145 (1.43)		-0.020*** (-6.04)		0.193*** (7.35)
Wage		0.041* (1.82)		0.401** (2.17)		0.106*** (9.82)		0.904*** (11.84)
Age		0.030 (1.53)		0.153 (1.01)		0.044*** (11.47)		0.411*** (13.29)
Fixed Effects	Provii	nce, SIC-3 In	dustry, Owr	n, Year		State, NAIC	S-6 Industry	
Noise Controls	Y	Y	Y	Y	Y	Y	Y	Y
R-squared # observations	0.41 3,233	0.43 3,123	0.40 2,236	0.43 1,935	0.26 32,000	0.27 32,000	0.33 13,000	0.37 13,000

Table 1. Export Status and Export Revenues

This table examines the relationship between firms' management practices, probability of exporting, and global export revenues. In Columns 1-2 and 5-6, the sample includes all Chinese firms and US establishments in the matched sample with balance sheet and management data, and the dependent variable is a binary indicator equal to 1 for exporters. In Columns 3-4 and 7-8, the sample includes all exporters in the matched sample with trade and management data, and the dependent variable is log total exports. *Management Score* is the standardized average score across all questions about firms' management practices. *Capital Intensity* is log net fixed asset per worker. *Wage* is log labor cost per employee. *Age* is log firm age in years. All columns control for the share of workers with a college degree; noise controls (interview duration and time of day; interviewer dummies; interviewee gender, reliability and competence as perceived by the interviewer). All regressions for China include fixed effects for firm province, main SIC-3 industry, year, and ownership status (private domestic, state-owned, foreign-owned). All regressions for the US include fixed effects for firm state and main NAICS-6 industry. Standard errors clustered by firm (China) and robust (US). US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

			China					SN		
Dep Variable:	Log # Dest	Log # Prod	Log # Dest-Prod	Log Avg Exports per Dest-Prod	Log Exports Top Dest- Prod	Log # Dest	Log # Prod	Log # Dest-Prod	Log Avg Exports per Dest-Prod	Log Exports Top Dest- Prod
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Management	0.185*** (2.80)	0.166*** (3.33)	0.215*** (2.89)	0.017 (0.20)	0.196* (1.74)	0.134*** (13.08)	0.165*** (15.32)	0.195*** (15.13)	0.177*** (12.75)	0.320*** (16.05)
Fixed Effects		Province	, SIC-3 Indust	try, Own, Yea	1		Sta	ite, NAICS-6 I	ndustry	
Noise Controls	<	< -<	< -<	< -<	< ~	<	<	< ~	< ~	< -<
R-squared	0.44 1 035	0.42 1 935	0.40 1 035	0.45 1 035	0.43 1 035	0.37	0.33	0.37 13 000	0.32	0.36 13 nnn
<u> </u>					:					<u>1</u>

Table 2. Extensive and Intensive Margins of Exports

a firm's highest-revenue destination-product in Columns 5 and 10. A product is HS 8-digit. All regressions for China include fixed effects for dependent variable is firms' log number of export destinations in Columns 1 and 6, log number of export products in Columns 2 and 7, log number of destination-product pairs in Columns 3 and 8, log average exports per destination-product in Columns 4 and 9, and log exports in NAICS-6 industry. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm firm province, main SIC-3 industry, year, and ownership status. All regressions for the US include fixed effects for firm state and main This table examines the relationship between firms' management practices and the extensive and intensive margins of their exports. The the 1%, 5%, and 10% level. (China) and robust (US). US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at

		Chii	na			US	6	
Dep Variable:	Log Export	Log Qual-Adj	Log Export	Log Export	Log Export	Log Qual-Adj	Log Export	Log Export
	Quality	Export Price	Price	Quantity	Quality	Export Price	Price	Quantity
Structural Parameter:	θ^{CH}	- δ^{CH}	θ^{CH} - δ^{CH}		θ^{us}	- δ^{US}	θ^{US} - δ^{US}	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Management	0.531*	-0.385*	0.146**	-0.200	0.048***	-0.045***	0.003	0.034***
	(1.95)	(-1.82)	(2.16)	(-1.49)	(2.60)	(-2.91)	(0.68)	(2.83)
Fixed Effects	Prov	/ince, Dest-Pro	duct, Own, `	Year		State, Dest	t-Product	
Noise Controls	Y	Y	Y	Y	Y	Y	Y	Y
Firm Controls	Y	Y	Y	Y	Y	Y	Y	Y
R-squared	0.92	0.89	0.92	0.79	0.96	0.95	0.97	0.83
# observations	58,101	58,101	58,101	58,101	290,000	290,000	290,000	290,000

Table 3. Production Efficiency and Product Quality

This table examines the relationship between firms' management practices and the price, quality, quality-adjusted price and quantity of their exports. The dependent variable is log export product quality in Columns 1 and 5, quality-adjusted log export unit value in Columns 2 and 6, log export unit value in Columns 3 and 7, and log export quantity in Columns 4 and 8, by firm-destination-product. Quality is estimated as demand elasticity (set to 5) x unit value + quantity as descrbed in the text. *Structural Parameter* is the model parameter identified from the reduced form coefficient on *Management*. A product is HS 8-digit. All regressions for China include fixed effects for firm province, destination-product pair, year, and ownership status. All regressions for the US include fixed effects for firm state and destination-product pair. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm. US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

			China					SN		
Dep Variable:	Log Imports	Log Imports Inputs	Log Avg Origin Income	Log Import Input Price	Log Import Input Quality	Log Imports	Log Imports Inputs	Log Avg Origin Income	Log Import Input Price	Log Import Input Quality
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Management	0.550***	0.222*	0.046**	0.101**	0.576***	0.344***	-0.003	0.037***	-0.001	0.051**
	(4.32)	(1.86)	(2.11)	(2.36)	(3.03)	(11.83)	(-0.03)	(3.89)	(-0.34)	(2.55)
Fixed Effects		Province, SIC	-3 Industry	, Own, Year			State, N	IAICS-6 Inc	dustry	
Noise Controls	~	×	×	×	×	×	×	×	×	×
Firm Controls	×	×	×	< ~	< ~	×	×	×	< ~	< -<
R-souared	0.56	0.50	0.38	0.81	0 78	0.31	0 27	0 21	76.0	0.03
# observations	1,778	1,778	1,778	76,626	76,626	10,000	10,000	10,000	140,000	140,000

Table 4. Imported Input Quality

imports in Columns 1 and 6, log share of imports in total intermediate inputs in Columns 2 and 7, log average GDP per capita across origin country-product in Columns 5 and 10. Quality is estimated as demand elasticity (set to 5) x unit value + quantity as described in the text. A countries in Columns 3 and 8, log import unit value by origin country-product in Columns 4 and 9, and log import product quality by origin pair fixed effects. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm in regressions for the US include fixed effects for firm state and main NAICS-6 industry. Columns 4-5 and 9-10 include origin country-product product is HS 8-digit. All regressions for China include fixed effects for firm province, main SIC-3 industry, year, and ownership status. All indicate significance at the 1%, 5%, and 10% level. Columns 1-5 and 9-10 and robust in Columns 6-8. US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and This table examines the relationship between firms' management practices and imported input quality. The dependent variable is log firm

		China			US	
Dep Variable:	Log # Origins	Log # Import Prod	Log # Origin-Prod	Log # Origins	Log # Import Prod	Log # Origin-Prod
	(1)	(2)	(3)	(4)	(5)	(6)
Management	0.168*** (4.24)	0.123* (1.82)	0.145** (2.09)	0.058*** (7.41)	0.079*** (6.81)	0.087*** (6.97)
Log # Export Products	0.245*** (7.69)	0.387*** (6.97)	0.441*** (7.77)	0.426*** (66.14)	0.561*** (58.70)	0.632*** (60.40)
Fixed Effects	Province,	SIC-3 Industry	, Own, Year	State	e, NAICS-6 Inc	dustry
Noise Controls	Y	Y	Y	Y	Y	Ý
Firm Controls	Y	Y	Y	Y	Y	Y
R-squared	0.61	0.64	0.67	0.56	0.51	0.53
# observations	1,566	1,566	1,566	10,000	10,000	10,000

Table 5. Assembly Complexity

This table examines the relationship between firms' management practices and imported input complexity. The dependent variable is firms' log number of origin countries in Columns 1 and 4, log number of imported products in Columns 2 and 5, and log number of origin country-product pairs in Columns 3 and 6. A product is HS 8-digit. All regressions for China include fixed effects for firm province, main SIC-3 industry, year, and ownership status. All regressions for the US include fixed effects for firm state and main NAICS-6 industry. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm (China) and robust (US). US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

Dep Variable:	Exporter Dummv	Log (1+ Exports)	Log Exports
	(1)	(2)	(3)
Panel A. Intention to Treat (Reduced Form)			
Treatment	0.189* (1.78)	0.665** (2.85)	0.416** (2.80)
Panel B. Management Impact (IV 2nd Stage))		
Management	0.899 (1.67)	3.16** (2.45)	1.95** (2.71)
1st Stage (Management on Treatment) F-test	35.8	35.8	20.9
Data frequency Years Firms Plants # Observations	Yearly 2008,11,14,17 17 31 109	Yearly 2008,11,14,17 17 31 109	Yearly 2008,11,14,17 12 22 66

Table 6. India RCT: Change in Managementand Trade Activity (2008-2017)

This table examines the relationship between firms' management practices and trade activity following a randomized control trial that provided management consulting to plants in the textile industry in India, 2008-2017. Results are at the plant-year level from the long-run follow-up in Bloom et al. (2017) collecting yearly data. The pre-treatment period is 2008, and the post-treatment period is 2011, 2014, and 2017. The sample includes 14 intervention plants in treated firms that received both initial diagnostics and management consulting, 6 non-intervention plants in treated firms that received only initial diagnostics, and 11 control plants that received neither. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level using the sample-size appropriate t-distribution tables.

		Export	Activity		Qua	lity and Efficie	ency	Imported	Input Qual	ity and Ass	sembly Cor	nplexity
Dep Variable:	Exporter Dummy	Log Exports	Log # Dest- Prod	Log Avg Exports per Dest-Prod	Log Avg Export Quality	Log Avg Qual-Adj Export Price	Log Avg Export Price	Log Imports	Log Avg Origin Income	Log Avg Imp Input Price	Log Avg Imp Input Quality	Log # Origin- Prod
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A. Lags:	Trade Activ	/ity 2011 a	nd Manag	jement 2010								
Management	0.029*** (9.48)	0.395*** (18.10)	0.208*** (16.19)	0.187*** (13.62)	0.053*** (3.25)	-0.059*** (-4.19)	-0.006 (-1.61)	0.374*** (13.23)	0.038*** (3.86)	-0.003 (-0.81)	0.045** (2.21)	-0.048*** (-2.62)
Fixed Effects Noise Controls Firm Controls	~ ~	\prec \prec	\prec \prec	\prec \prec	\prec \prec	State, NAICS Y Y	γ γ γ	\prec \prec	~ ~	~ ~	~ ~	\prec \prec
R-squared # observations	0.29 31,000	0.39 13,000	0.33 13,000	0.32 13,000	0.97 13,000	0.96 13,000	0.98 13,000	0.33 10,000	0.21 10,000	0.97 10,000	0.93 10,000	0.91 10,000
Panel B. First [Differences	: Δ Trade A	Activity 20	05→2010 an	d Δ Manage	∍ment 2005→	2010					
Management	0.004*** (3.19)	0.055*** (4.12)	0.031*** (4.28)	0.025** (2.53)	0.024** (2.25)	-0.024** (-2.49)	0.001 (0.41)	0.050*** (2.76)	-0.018*** (-2.88)	-0.001 (-0.53)	0.057*** (4.48)	0.031*** (3.69)
Fixed Effects Noise Controls Firm Controls	~ ~	\prec \prec	\prec \prec	\prec \prec	\prec \prec	State, NAICS Y Y	γ γ γ	\prec \prec	~ ~	~ ~	~ ~	\prec \prec
R-squared # observations	0.10 31,000	0.06 13,000	0.07 13,000	0.06 13,000	0.042 13,000	0.04 13,000	0.08 13,000	0.09 10,000	0.09 10,000	0.07 10,000	0.07 10,000	0.08 10,000
This table exam variables are de	ines the rela fined in Tat	ationship be bles 1-5. In nt variables	etween firr Panel A, t are within	ns' managem he dependent -firm changes	ent practice variable is from 2005	s, export and i for year 2011, to 2010. All re	mport activit while the ma	y in the pane anagement v	el for US firr ∕ariable is fo controls. fix	ns with mat or year 2010 ed effects fi	ched data ir). In Panel E or firm state	1 2010. All }, both the and main

Table 7. US Panel: Management and Trade Activity over Time

statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

Figure 1A. US Management and Organizational Practices Survey: Examples

Example 1: Monitoring

2 In 2005 and 2010, how many key performance indicators were monitored at this establish	ment?	
Examples: Metrics on production, cost, waste, quality, inventory, energy, absenteeism an	d deliveries on	time.
Check one box for each year	2005	2010
1-2 key performance indicators		
3-9 key performance indicators		
10 or more key performance indicators		
No key performance indicators (If no key performance indicators in both years, SKIP to 6)		

Example 2: Targets

8 In 2005 and 2010, who was aware of the production targets at this establishment? Check on	e box for eac	h year
	2005	2010
Only senior managers		
Most managers and some production workers		
Most managers and most production workers		
All managers and most production workers		

Example 3: Incentives

In 2005 and 2010, what was the primary way managers were promoted at this establish	iment?	
Check one box for each year	2005	2010
Promotions were based solely on performance and ability		
Promotions were based partly on performance and ability, and partly on other factors (for example, tenure or family connections)		
Promotions were based mainly on factors other than performance and ability (for example, tenure or family connections)		
Managers are normally not promoted		

This figure provides examples of the 16 questions in the MOPS survey for the US that span the management of physical capital resources (subdivided into monitoring production and setting targets) and of human capital resources (incentives design).

Figure 1B. China World Management Survey: Examples

Example 1: Monitoring: How is performance tracked?

Score	 (1): Measures tracked do not indicate directly if overall business objectives are being met. Certain processes aren't tracked at all 	(3): Most key performance indicators are tracked formally. Tracking is overseen by senior management	(5): Performance is continuously tracked and communicated, both formally and informally, to all staff using a range of visual management tools
-------	---	--	--

Example 2: Targets: How are targets set?

Score	(1): Goals are exclusively financial or operational	(3): Goals include non- financial targets, which form part of the performance appraisal of top management only	(5): Goals are a balance of financial and non-financial targets. Senior managers believe the non-financial targets are often more inspiring and challenging than
		only	challenging than financials alone

Example 3: Incentives: How does promotion work?

Score	(1) People are promoted primarily upon the basis of tenure, irrespective of performance (ability & effort)	(3) People are promoted primarily upon the basis of performance	(5) We actively identify, develop and promote our top performers
-------	---	--	---

This figure provides examples of the 18 questions in the WMS survey for China that span the management of physical capital resources (subdivided into monitoring production and setting targets) and of human capital resources (incentives design).











This figure plots the MOPS management score distribution for the US (top) and the WMS ma score distribution for China (bottom). The management scores are averaged across all questic being normalized for the regression analysis.



Figure 3. Average WMS Management Practices across Countries

This figure plots the WMS average management score across all firms in a country, averaged over all WMS waves from 2004 to 2014. Each firm is scored on 18 questions and each question is marked on a scale of 1 to 5, such that the overall firm and country scores have a range of 1 to 5.



Figure 4. India RCT: Change in TFP vs. Change in Management (2008-2011)

This figure displays the relationship between the improvement in firms' management practices and total factor productivity (in logs) following a randomized control trial that provided management consulting to plants in the textile industry in India, 2008-2011. It plots the firm-by-week change in log total factor productivity against the firm-by-week change in the management score, both relative to their pre-experiment average. See Bloom et al. (2013) for experiment details.





This figure displays the relationship between the improvement in firms' management practices and quality control in production following a randomized control trial that provided management consulting to plants in the textile industry in India, 2008-2011. It plots the firm-by-week change in the log quality defects index against the firm-by-week change in the management score, both relative to their pre-experiment average. The quality defects index measures the severity-weighted number of defects per roll of fabric. See Bloom et al. (2013) for experiment details.

Appendix Table 1. Summary Statistics

	С	hina		US
	Exporters	Non-exporters	Exporters	Non-exporters
# Observations	1,875	1,358	14,000	18,000
Management	0.06	-0.09	0.12	-0.26
Log Gross Output	11.72	11.55	10.6	9.55
Log Employment	6.46	6.15	4.76	3.96
TFPR	4.86	4.77	4.3	4.07
Log Value Added / L	3.73	3.95	5.04	4.78

Panel A. Characteristics of exporters and non-exporters

Panel B. Firms' management, export and import activity

	CI	nina	L	JS
	Mean	St Dev	Mean	St Dev
Management	0	1	0	1
# Export Observations	2,236		13,000	
Log Exports	14.80	2.31	13.79	2.77
# Export Products	8.65	11.58	18.94	47.50
# Export Destinations	12.85	14.99	12.95	16.72
# Import Observations	2,048		10,000	
Log Imports	13.87	2.97	13.93	2.96
# Import Products	33.45	51.43	19.67	43.09
# Import Origins	6.30	5.67	6.20	8.02

This tables provides summary statistics. China: all firms in the matched WMS-ASIE sample for 1999-2007 (Panel A) and all exporters in the matched WMS-CCTS sample for 2000-2008 (Panel B). US: all plants in the matched MOPS-ASM sample for 2010 (Panel A) and all exporting firms in the matched MOPS-LFTTD sample for 2010 (Panel B).

		China			US	
Dep Variable:		Log Profits			Log Profits	
	(1)	(2)	(3)	(4)	(5)	(6)
Management	0.546*** (6.98)	0.387*** (5.70)	0.361*** (5.43)	0.431*** (32.61)	0.340*** (27.01)	0.111*** (10.21)
Log Dom Sales			0.097*** (5.85)			0.671*** (64.28)
Fixed Effects	Province, S	IC-3 Industry	, Own, Year	State	, NAICS-6 In	dustry
Noise Controls	Y	Y	Y	Y	Y	Y
Firm Controls	Ν	Y	Y	Ν	Y	Y
R-squared # observations	0.45 2,520	0.55 2,438	0.57 2,438	0.71 13,000	0.75 13,000	0.85 13,000

Appendix Table 2. (Export) Profits

This table examines the relationship between firms' management practices and profits. The dependent variable is firms' log profits. All regressions for China include noise controls and fixed effects for firm province, main SIC-3 industry, year, and ownership status. All regressions for the US include noise controls and fixed effects for firm state and main NAICS-6 industry. Columns 2-3 and 5-6 also include a full set of firm controls as described in Table 1. Standard errors clustered by firm (China) and robust (US). US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

			Chin	۵				SN		
Dep Variable:	Log # Dest	Log # Prod	Log # Dest-Pro	Log Avg Exports per Dest-Prod	Log Exports Top Dest- Prod	Log # Dest	Log # Prod	Log # Dest-Prod	Log Avg Exports per Dest-Prod	Log Exports Top Dest- Prod
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Management	0.159**	0.152***	0.200***	0.062	0.226**	0.179***	0.213***	0.257***	0.231***	0.418***
	(2.51)	(3.06)	(2.72)	(0.75)	(2.11)	(17.44)	(19.67)	(19.76)	(16.62)	(20.8)
Fixed Effects		Province	, SIC-3 Indu	ıstry, Own, Yea	7		Sta	ate, NAICS-6 I	Industry	
Noise Controls	~	~	×	×	~	×	~	×	×	×
Firm Controls	z	z	z	z	z	z	z	z	z	z
R-squared	0.43	0.41	0.40	0.38	0.39	0.33	0.29	0.33	0.28	0.32
# observations	2,236	2,236	2,236	2,236	2,236	13,000	13,000	13,000	13,000	13,000
This table exam	nines the r	elationship	between fi	rms' managem	ent practices :	and the ext	ensive and	intensive ma	argins of their	r exports. All
This table exam	nines the r	elationship	between fil	rms' managem	ent practices a	and the exte	ensive and	intensive ma	argins of their	r exports. All

Appendix Table 3. Extensive and Intensive Margins of Exports: No Firm Controls

variables, fixed effects, and noise controls are as described in Table 2, but the regressions exclude firm controls. Standard errors clustered by firm (China) and robust (US). US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

			China					SN		
Dep Variable:	Log Dom Sales	Exporter Dummy	Log Exports	Log # Dest-Prod	Log Avg Exports per Dest-Prod	Log Dom Sales	Exporter Dummy	Log Exports	Log # Dest-Prod	Log Avg Exports per Dest-Prod
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Management	0.475***	0.058***	0.250*	0.219***	0.032	0.344***	0.022***	0.164***	0.072***	0.092***
	(2.97)	(3.32)	(1.96)	(2.96)	(0.37)	(29.43)	(6.92)	(7.35)	(5.54)	(6.46)
Log Dom Sales		-0.025***	-0.035	-0.007	-0.028		0.028***	0.605***	0.358***	0.247***
		()	()	(0:10)	()		10:01	(00:0-)	(00:00)	(1.00)
Fixed Effects		Province, S	IC-3 Indust	try, Own, Yea	ar		State	, NAICS-6 I	ndustry	
Noise Controls	×	×	×	×	×	×	×	×	×	×
Firm Controls	×	×	×	×	×	×	×	×	×	×
R-squared	0.98	0.43	0.44	0.40	0.45	0.49	0.27	0.45	0.43	0.35
# observations	1,935	3,123	1,935	1,935	1,935	13,000	32,000	13,000	13,000	13,000
Thic table evan	ainon tho m	Nationatin h	otwoon fire		mont prontinon	domostio	and oxport		dopoodopt v	winklos are

Appendix Table 4. Export vs. Domestic Activity

controls as described in Table 1. Standard errors clustered by firm (China) and robust (US). US sample sizes rounded for disclosure regressions for the US include fixed effects for firm state and main NAICS-6 industry. All columns also include a full set of firm and noise defined in Tables 1-2. All regressions for China include fixed effects for firm province, main SIC-3 industry, year, and ownership status. All reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level. inis table examines the relationship between times management practices, domestic and export activity. All dependent variables are

Dep Variable:	TFPR	Exporter Dummy	Log Exports	Exporter Dummy	Log Exports	Log Export Quality	Log Qual-Adj Exp Price	Log Imp Input Quality
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A. China								
Management	0.150*** (3.48)			0.053*** (2.93)	0.287** (2.34)	0.520* (1.89)	-0.363* (-1.69)	0.592*** (3.14)
TFPR		-0.006 (-0.45)	0.274*** (3.54)					
Non-Management TI	=PR			-0.006 (-0.49)	0.246*** (3.28)	0.242** (2.30)	-0.192** (-2.32)	0.411*** (2.87)
Fixed Effects Noise, Firm Controls Country-Product FE	Y 	Y 	Provi Y 	nce, SIC-3 Y 	Industry, O Y 	wn, Year Y Y	Y Y	Y Y
R-squared # observations	0.49 2,800	0.42 2,802	0.44 1,880	0.43 2,800	0.46 1,880	0.90 54,565	0.89 54,565	0.78 70,270
Panel B. US								
Management	0.090*** (10.10)			0.026*** (8.66)	0.358*** (16.37)	0.041*** (2.96)	-0.045*** (-3.64)	0.049** (2.50)
TFPR		0.040*** (11.49)	0.307*** (12.09)					
Non-Management TI	-PR			0.037*** (10.50)	0.273*** (11.12)	0.025** (2.30)	-0.024** (-2.38)	0.035*** (2.58)
Fixed Effects Noise, Firm Controls	Y	Y	Y	State, NAI Y	CS-6 Indus Y	stry Y	Y	Y
R-squared # observations	 0.83 32,000	 0.28 32,000	 0.38 13,000	 0.28 32,000	 0.41 13,000	т 0.97 290,000	r 0.96 290,000	r 0.93 140,000

Appendix Table 5. Management vs. TFPR

This table examines the relationship between firms' management practices, total factor productivity, and trade activity. All dependent variables are defined in Tables 1, 3 and 4. *TFPR* is revenue-based TFP measured as in Levinsohn-Petrin. *Non-Management TFPR* is the residual from the regression of TFPR on management and no other controls or fixed effects. All regressions for China include fixed effects for firm province, main SIC-3 industry, year, and ownership status. All regressions for the US include fixed effects for firm state and main NAICS-6 industry. Columns 6-8 include country-product pair fixed effects. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm, except for Columns 1-5 for the US where they are robust. US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

Appendix Table 6. Production Efficiency and Product Quality: Robustness

	-							
		Chi	na			US	6	
Dep Variable:	Log Export Quality	Log Qual-Adj Export Price	Log Export Price	Log Export Quantity	Log Export Quality	Log Qual-Adj Export Price	Log Export Price	Log Export Quantity
Structural parameter(s)	θ^{CH}	- δ^{CH}	θ^{CH} - δ^{CH}		θ^{US}	- δ^{US}	$\theta^{\text{US}} \text{-} \delta^{\text{US}}$	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Management	0.332* (1.96)	-0.185 (-1.48)	0.149** (2.16)	-0.203 (-1.49)	0.042*** (2.99)	-0.046*** (-3.68)	0.003 (0.68)	0.034*** (2.83)
Fixed Effects	Province, Dest-Product, Own, Year				State, Dest-Product			
Noise Controls Firm Controls	Y Y	Y Y	Y Y	Y Y	Y Y	Y Y	Y Y	Y Y
R-squared # observations	0.90 58,101	0.89 58,101	0.92 58,101	0.79 58,101	0.97 290,000	0.96 290,000	0.97 290,000	0.83 290,000

Panel A. Sector-specific demand elasticity (Broda-Weinstein)

Panel B. Controlling for market power

		Chii	าล			US	3	
Dep Variable:	Log Export Quality	Log Qual-Adj Export Price	Log Export Price	Log Export Quantity	Log Export Quality	Log Qual-Adj Export Price	Log Export Price	Log Export Quantity
Structural parameter(s)	θ^{CH}	- δ ^{CH}	θ^{CH} - δ^{CH}		θ^{us}	- δ^{US}	$\theta^{\text{US}} \text{-} \delta^{\text{US}}$	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Management	0.531* (1.95)	-0.385* (-1.82)	0.146** (2.16)	-0.200 (-1.49)	0.040** (2.36)	-0.044*** (-2.99)	-0.004 (-1.01)	0.058*** (4.69)
Market Share	0.001 (0.94)	-0.001 (-1.01)	0.000 (0.28)	0.001 (1.43)	0.038* (1.78)	-0.040* (-1.78)	-0.002* (-1.72)	0.048* (1.78)
Fixed Effects	Province, Dest-Product, Own, Year				State, Dest-Product			
Noise Controls	Y	Y	Y	Y	Y	Y	Y	Y
Firm Controls	Y	Y	Y	Y	Y	Y	Y	Y
R-squared # observations	0.90 58,101	0.89 58,101	0.92 58,101	0.79 58,101	0.97 290,000	0.96 290,000	0.98 290,000	0.84 290,000

This table examines the robustness of the relationship between firms' management practices and the price, quality, quality-adjusted price and quantity of their exports. All variables, controls and fixed effects are as described in Table 3 with two exceptions. In Panel A, quality and quality-adjusted prices are constructed using industry-specific demand elasticities from Broda-Weinstein (2006). In Panel B, an additional control is added for a firm's market power: its share of total Chinese exports by destination-product-year (Columns 1-4) or of total US exports by destination-product (Columns 5-8). Standard errors clustered by firm. US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% level.

	Export /	Activity	Qua	lity and Effic	iency	Importec Assen	l Input Qua hbly Compl	lity and exity
Dep Variable:	Exporter Dummy	Log Exports	Log Export Quality	Log Qual-Adj Export Price	Log Export Price	Log Avg Origin Income	Log Imp Input Quality	Log # Origin- Prod
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A. China								
Monitoring & Targets	0.061*** (2.68)	0.012 (0.08)	0.558* (1.84)	-0.378 (-1.58)	0.180** (2.54)	0.059** (2.19)	0.353 (1.64)	0.373*** (3.89)
Incentives	-0.030 (-0.58)	0.266* (1.96)	-0.008 (-0.03)	-0.024 (-0.11)	-0.032 (-0.52)	-0.013 (-0.42)	0.289 (1.23)	-0.195** (-2.09)
Fixed Effects	<	<	< Pro	vince, SIC-3 Ir V	ndustry, Own, \ ע	rear V	<	<
Country-Product FE	¦ →	-	≺ -	×	≺ -	-		¦ →
R-squared # observations	0.43 3,123	0.43 1,935	0.9 58,101	0.89 58,101	0.92 58,101	0.38 1,778	0.78 76,626	0.61 1,778
Panel B. US								
Monitoring & Targets	0.022*** (6.99)	0.307*** (13.11)	0.050** (2.56)	-0.050*** (-3.88)	-0.005 (-1.10)	0.045*** (4.52)	0.052** (2.57)	0.101*** (7.67)
Incentives	0.013*** (4.63)	0.141*** (6.57)	0.017 (1.03)	-0.006 (-0.057)	0.001 (0.16)	-0.003 (-0.29)	0.014 (0.86)	0.011 (0.88)
Fixed Effects Noise, Firm Controls	~	×	~	State, NAIC Y	:S-6 Industry Υ	~	~	~
Country-Product FE R-squared	 0.27	0.39	0.96	0.96	Y 0.97	0.21	0.93	
# observations	32.000	13.000	290.000	290.000	290.000	10.000	140.000	10.000

Appendix Table 7. Management Components

industry, year, and ownership status. All regressions for the US include fixed effects for firm state and main NAICS-6 industry. Columns 3-5 and 7 include country-product pair fixed effects. All columns also include a full set of firm and noise controls as described in Table 1. Standard errors clustered by firm, except for Columns 1, 2, 6 and 8 for the US This table examines the role of different components of firms' management practices as described in the text. All variables are defined in Tables 1-5. All regressions for China include fixed effects for firm province, main SIC-3 indicate significance at the 1%, 5%, and 10% level. where they are robust. US sample sizes rounded for disclosure reasons. T-statistics in parentheses. ***, **, and *

Managing Trade: Evidence from China and the US^*

Nick Bloom Stanford University Kalina Manova University College London

John Van Reenen *MIT* Stephen Teng Sun Peking University

Zhihong Yu University of Nottingham

January 2018

Abstract

This appendix provides formal proofs for the baseline model in the paper and presents three theoretical extensions for endogenous choice of management practices, multiple components of firm ability, and endogenous choice of input and output quality.

1 Proofs for Baseline Model

1.1 Set Up

Product demand. The representative consumer in country j has CES utility

$$U_j = \left[\int_{i \in \Omega_j} (q_{ji} x_{ji})^{\alpha} di \right]^{\frac{1}{\alpha}}$$
(1.1)

^{*}We thank Rui Zhang and Xincheng Qiu of Peking University for his excellent research assistance for this appendix.

where q_{ji} and x_{ji} are quality and quantity consumed by country j of variety i, and Ω_j is the set of goods available to j. The elasticity of substitution across products is $\sigma = 1/(1-\alpha) > 1$ with $0 < \alpha < 1$. If total expenditure in country j is R_j , j's demand for variety i is

$$x_{ji} = R_j P_j^{\sigma - 1} q_{ji}^{\sigma - 1} p_{ji}^{-\sigma}.$$
 (1.2)

Proof. The utility maximization problem is

$$\max_{\{x_{ji}\}} U_j = \left[\int_{i \in \Omega_j} (q_{ji} x_{ji})^{\alpha} di \right]^{\frac{1}{\alpha}} \quad \text{s.t.} \quad \int_{i \in \Omega_j} (p_{ji} x_{ji}) di = R_j.$$
(1.3)

where p_{ji} is the price of variety *i* in country *j*. Define the Lagrangian function as

$$L = \left[\int_{i \in \Omega_j} (q_{ji} x_{ji})^{\alpha} di \right]^{\frac{1}{\alpha}} + \lambda \quad R_j - \int_{i \in \Omega_j} (p_{ji} x_{ji}) di \right).$$
(1.4)

The first order condition implies:

$$\frac{\partial L}{\partial x_{ji}} = \left[\int_{i \in \Omega_j} (q_{ji} x_{ji})^{\alpha} di \right]^{\frac{1-\alpha}{\alpha}} (q_{ji} x_{ji})^{\alpha-1} q_{ji} - \lambda p_{ji} = 0, \qquad (1.5)$$

$$\implies x_{ji} = \frac{\left(\lambda \frac{p_{ji}}{q_{ji}}\right)^{\frac{1}{\alpha-1}} \left[\int_{i \in \Omega_j} \left(q_{ji} x_{ji}\right)^{\alpha} di\right]^{\frac{1}{\alpha}}}{q_{ji}}.$$
(1.6)

Substituting for x_{ji} in the budget constraint and rearranging yields

$$\lambda = \left\{ \frac{\left[\int_{i \in \Omega_j} (q_{ji} x_{ji})^{\alpha} di \right]^{\frac{1}{\alpha}} \int_{i \in \Omega_j} \left(\frac{p_{ji}}{q_{ji}} \right)^{1-\sigma} di}{R_j} \right\}^{1-\alpha} \text{ and } (1.7)$$

$$x_{ji} = R_j P_j^{\sigma-1} q_{ji}^{\sigma-1} p_{ji}^{-\sigma}, (1.8)$$

where we have used $\sigma = 1/(1-\alpha)$ and defined $P_j \equiv \left[\int_{i\in\Omega_j} \left(\frac{p_{ji}}{q_{ji}}\right)^{1-\sigma} di\right]^{\frac{1}{1-\sigma}}$ as a quality-adjusted ideal price index.

1.2 Profit Maximization

Optimal firm behavior. Individual producers separately maximize profits for each destinationproduct market by solving

$$\max_{p_{ji}, x_{ji}} \pi_{ji} = p_{ji} x_{ji} - \tau_j x_{ji} \left(\varphi \lambda_i\right)^{\theta - \delta} - f_{pj}$$
(1.9)
s.t. $x_{ji} = R_j P_j^{\sigma - 1} q_{ji}^{\sigma - 1} p_{ji}^{-\sigma}.$

Product quality is exogenously determined by the quality production function as $q_{ji} = q_i = (\varphi \lambda_i)^{\theta}$. A producer with management competence φ and product expertise λ_i will therefore charge a constant mark-up $\frac{1}{\alpha}$ over marginal cost and have the following price, quantity, quality, quality-adjusted price, revenues and profits for product *i* in market *j*:

$$p_{ji}(\varphi,\lambda_i) = \frac{\tau_j(\varphi\lambda_i)^{\theta-\delta}}{\alpha}, \qquad x_{ji}(\varphi,\lambda_i) = R_j P_j^{\sigma-1} \left(\frac{\alpha}{\tau_j}\right)^{\sigma} (\varphi\lambda_i)^{\delta\sigma-\theta}, \qquad (1.10)$$

$$q_i(\varphi,\lambda_i) = (\varphi\lambda_i)^{\theta}, \quad p_{ji}(\varphi,\lambda_i)/q_i(\varphi,\lambda_i) = \frac{\tau_j(\varphi\lambda_i)^{-\delta}}{\alpha},$$
 (1.11)

$$r_{ji}(\varphi,\lambda_i) = R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1} (\varphi\lambda_i)^{\delta(\sigma-1)}, \qquad \pi_{ji}(\varphi,\lambda_i) = \frac{r_{ji}(\varphi,\lambda_i)}{\sigma} - f_{pj}.$$
(1.12)

Proof. Define the Lagrangian function as

$$L = p_{ji}x_{ji} - \tau_j x_{ji} \left(\varphi\lambda_i\right)^{\theta-\delta} - f_{pj} + \mu \left(R_j P_j^{\sigma-1} \left(\varphi\lambda_i\right)^{\theta(\sigma-1)} p_{ji}^{-\sigma} - x_{ji}\right).$$
(1.13)

The first order conditions are:

$$\frac{\partial L}{\partial x_{ji}} = p_{ji} - \tau_j \left(\varphi \lambda_i\right)^{\theta - \delta} - \mu = 0, \qquad (1.14)$$

$$\frac{\partial L}{\partial p_{ji}} = x_{ji} - \sigma \mu R_j P_j^{\sigma-1} \left(\varphi \lambda_i\right)^{\theta(\sigma-1)} p_{ji}^{-\sigma-1} = 0, \qquad (1.15)$$

$$\frac{\partial L}{\partial \mu} = R_j P_j^{\sigma-1} \left(\varphi \lambda_i\right)^{\theta(\sigma-1)} p_{ji}^{-\sigma} - x_{ji} = 0.$$
(1.16)

Plugging the second condition into the third one, one obtains $p_{ji} = \sigma \mu$. Substituting into the first condition, it follows that $\mu = \tau_j (\varphi \lambda_i)^{\theta - \delta} / (\sigma - 1)$. Using simple algebra and $\sigma = 1/(1 - \alpha)$ delivers the following expressions for the outcomes of interest:

$$p_{ji}(\varphi,\lambda_i) = \sigma \mu = \sigma \frac{\tau_j(\varphi\lambda_i)^{\theta-\delta}}{\sigma-1} = \frac{\tau_j(\varphi\lambda_i)^{\theta-\delta}}{\alpha}, \qquad (1.17)$$

$$x_{ji}(\varphi,\lambda_i) = R_j P_j^{\sigma-1}(\varphi\lambda_i)^{\theta(\sigma-1)} p_{ji}^{-\sigma} = R_j P_j^{\sigma-1} \left(\frac{\alpha}{\tau_j}\right)^{\sigma} (\varphi\lambda_i)^{\delta\sigma-\theta}, \qquad (1.18)$$

$$\frac{p_{ji}(\varphi,\lambda_i)}{q_i(\varphi,\lambda_i)} = \frac{\tau_j(\varphi\lambda_i)^{-\delta}}{\alpha}, \qquad (1.19)$$

$$r_{ji}(\varphi,\lambda_i) = p_{ji}x_{ji} = R_j \left(\frac{P_j\alpha}{\tau_j}\right)^{\sigma-1} (\varphi\lambda_i)^{\delta(\sigma-1)}, \qquad (1.20)$$

$$\pi_{ji}(\varphi,\lambda_i) = p_{ji}x_{ji} - \tau_j x_{ji}(\varphi\lambda_i)^{\theta-\delta} - f_{pj} = (1-\alpha)r_{ji} - f_{pj} = \frac{r_{ji}(\varphi,\lambda_i)}{\sigma} - f_{pj}.(1.21)$$

1.3 Selection into Products and Markets

=

Product expertise cut-off for production. Since profits $\pi_d(\varphi, \lambda_i)$ increase with product expertise λ_i , there is a zero-profit expertise level $\lambda^*(\varphi)$ for each management ability draw φ below which the firm will not produce *i* for the domestic market. This cut-off is defined by the zero-profit condition $\pi_d(\varphi, \lambda^*(\varphi)) = 0$ and is decreasing in φ , i.e. $\frac{d\lambda^*(\varphi)}{d\varphi} < 0$.

Proof. The definition of the product expertise cut-off $\lambda^*(\varphi)$ delivers a closed-form solution for it:

$$\pi_d(\varphi,\lambda^*(\varphi)) = 0 \Leftrightarrow r_d(\varphi,\lambda^*(\varphi)) = R_d(P_d\alpha)^{\sigma-1}(\varphi\lambda^*(\varphi))^{\delta(\sigma-1)} = \sigma f_p$$
(1.22)

$$\Rightarrow \lambda^*(\varphi) = \frac{1}{\varphi} \left[\frac{\sigma f_p}{R_d \left(P_d \alpha \right)^{\sigma - 1}} \right]^{\frac{1}{\delta(\sigma - 1)}}.$$
 (1.23)

Therefore $\frac{d\lambda^*(\varphi)}{d\varphi} < 0.$

Product expertise cut-off for exporting. Similarly, export profits $\pi_{ji}(\varphi, \lambda_i)$ increase with product expertise λ_i , such that there is a cut-off expertise level $\lambda_j^*(\varphi)$ for each management ability draw φ below which the firm will not export product *i* to country *j*. This cut-off is defined by the zero-profit condition $\pi_{ji}(\varphi, \lambda_j^*(\varphi)) = 0$ and is decreasing in φ , i.e. $\frac{d\lambda_j^*(\varphi)}{d\varphi} < 0.$

Proof. The definition of the export product expertise cut-off $\lambda_j^*(\varphi)$ delivers a closed-form solution for it:

$$\pi_{ji}\left(\varphi,\lambda_{j}^{*}\left(\varphi\right)\right) = 0 \Leftrightarrow r_{ji}\left(\varphi,\lambda_{j}^{*}\left(\varphi\right)\right) = R_{j}\left(\frac{P_{j}\alpha}{\tau_{j}}\right)^{\sigma-1}\left(\varphi\lambda_{j}^{*}\left(\varphi\right)\right)^{\delta(\sigma-1)} = \sigma f_{pj} \qquad (1.24)$$

$$\implies \lambda_j^*(\varphi) = \frac{1}{\varphi} \left[\frac{\sigma f_{pj}}{R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1}} \right]^{\frac{1}{\delta(\sigma-1)}}.$$
(1.25)

Therefore $\frac{d\lambda_j^*(\varphi)}{d\varphi} < 0.$

Management ability cut-off for exporting. The export profits in country j of a firm with management competence φ are:

$$\pi_{j}(\varphi) = \int_{\lambda_{j}^{*}(\varphi)}^{\infty} \pi_{ji}(\varphi, \lambda) z(\lambda) d\lambda - f_{hj}.$$
(1.26)

Since export profits $\pi_j(\varphi)$ increase with management ability φ , only firms with management level above a cut-off φ_j^* will service destination j. This cut-off is defined by the zero-profit condition $\pi_j(\varphi_j^*) = 0$.

Proof. According to Leibniz's rule,

$$\frac{d\pi_{j}\left(\varphi\right)}{d\varphi} = \int_{\lambda_{j}^{*}(\varphi)}^{\infty} \frac{\partial\pi_{ji}\left(\varphi,\lambda\right)}{\partial\varphi} z\left(\lambda\right) d\lambda - \pi_{ji}\left(\varphi,\lambda_{j}^{*}\left(\varphi\right)\right) z\left(\lambda_{j}^{*}\left(\varphi\right)\right) \frac{d\lambda_{j}^{*}\left(\varphi\right)}{d\varphi}.$$
(1.27)

Since $r_{ji}(\varphi, \lambda_i) = R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1} (\varphi \lambda_i)^{\delta(\sigma-1)}$ and $\pi_{ji}(\varphi, \lambda_i) = \frac{r_{ji}(\varphi, \lambda_i)}{\sigma} - f_{pj}$, it follows that

$$\frac{\partial \pi_{ji}(\varphi,\lambda)}{\partial \varphi} = \frac{1}{\sigma} \frac{\partial r_{ji}(\varphi,\lambda)}{\partial \varphi} = \frac{\delta(\sigma-1)}{\sigma} R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1} (\varphi \lambda)^{\delta(\sigma-1)-1} \lambda > 0$$
(1.28)

because $\delta > 0$ and $\sigma > 1$. We have already proved that $\frac{d\lambda_j^*(\varphi)}{d\varphi} < 0$. Therefore $\frac{d\pi_j(\varphi)}{d\varphi} > 0$, such that export profits in country j increase with management ability and only firms above a zero-profit management cut-off will export to j.

Management ability cut-off for production. Firm φ 's global profits are given by

$$\pi(\varphi) = \pi_d(\varphi) + \sum_j \pi_j(\varphi) = \int_{\lambda^*(\varphi)}^{\infty} \pi_d(\varphi, \lambda) \, z(\lambda) \, d\lambda + \sum_j \left(\int_{\lambda^*_j(\varphi)}^{\infty} \pi_{ji}(\varphi, \lambda) \, z(\lambda) \, d\lambda - f_{hj} \right) - f_h$$
(1.29)

Since global profits $\pi(\varphi)$ increase with management ability φ , firms with management below a minimum level φ^* will be unable to break even and exit immediately upon learning their attributes. This cut-off is defined by the zero-profit condition $\pi(\varphi^*) = 0$.

Proof. According to Leibniz's rule,

$$\frac{d\pi\left(\varphi\right)}{d\varphi} = \int_{\lambda^{*}(\varphi)}^{\infty} \frac{\partial\pi_{d}\left(\varphi,\lambda\right)}{\partial\varphi} z\left(\lambda\right) d\lambda - \pi_{d}\left(\varphi,\lambda^{*}\left(\varphi\right)\right) z\left(\lambda^{*}\left(\varphi\right)\right) \frac{d\lambda^{*}\left(\varphi\right)}{d\varphi} + \sum_{j} \frac{d\pi_{j}\left(\varphi\right)}{d\varphi}.$$
 (1.30)

Since $r_d(\varphi, \lambda_i) = R_d (P_d \alpha)^{\sigma-1} (\varphi \lambda_i)^{\delta(\sigma-1)}$ and $\pi_d(\varphi, \lambda_i) = \frac{r_d(\varphi, \lambda_i)}{\sigma} - f_p$, it follows that

$$\frac{\partial \pi_d \left(\varphi, \lambda\right)}{\partial \varphi} = \frac{1}{\sigma} \frac{\partial r_d \left(\varphi, \lambda\right)}{\partial \varphi} = \frac{\delta(\sigma - 1)}{\sigma} R_d \left(P_d \alpha\right)^{\sigma - 1} \left(\varphi \lambda\right)^{\delta(\sigma - 1) - 1} \lambda > 0 \tag{1.31}$$

because $\delta > 0$ and $\sigma > 1$. We have already proved that $\frac{d\lambda^*(\varphi)}{d\varphi} < 0$ and $\frac{d\pi_j(\varphi)}{d\varphi} > 0$. Therefore $\frac{d\pi(\varphi)}{d\varphi} > 0$, such that global profits increase with management ability and only firms above a zero-profit management cut-off will commence production.

1.4 Empirical Predictions

Proposition 1. Better managed firms are more likely to export.

Proof. This proposition follows from the result that total export profits $\pi_X(\varphi) = \sum_j \pi_j(\varphi)$ increase with management ability φ . On the intensive margin, we have already established that bilateral export profits increase with management competence, $\frac{\partial \pi_j(\varphi)}{\partial \varphi} > 0$. On the extensive margin, only firms with ability $\varphi \geq \varphi_j^*$ will sell to destination j. For destinations $j = \{1, 2, ..., J\}$, denote

$$\varphi_X^* = \min \left\{ \varphi_1^*, \varphi_2^*, ..., \varphi_J^* \right\}$$
(1.32)

Since firms with higher φ are more likely to have both $\varphi \ge \varphi_j^*$ for any j and $\varphi \ge \varphi_X^*$ overall, they have a higher propensity to export to any given destination j, as well as a higher propensity to be exporters, i.e. to export to at least one destination. The proof to the next proposition is closely related and provides detailed derivations for these claims.

Proposition 2. Better managed firms export more products to more destination markets and earn higher export revenues and profits.

Proof. First, denote the number of destinations a firm enters as $n(\varphi) = \sum_{j} I(\varphi \ge \varphi_{j}^{*})$, where

$$I\left(\varphi \ge \varphi_j^*\right) = \begin{cases} 1, & \varphi \ge \varphi_j^*\\ 0, & \varphi < \varphi_j^* \end{cases}$$
(1.33)

A higher φ means that a larger number of destinations j satisfy $\varphi \ge \varphi_j^*$ because $\frac{\partial I(\varphi \ge \varphi_j^*)}{\partial \varphi} > 0$. Therefore $n(\varphi)$ is increasing in φ and better managed exporters enter more markets, i.e. $\frac{\partial n(\varphi)}{\partial \varphi} > 0$.

 $\frac{\partial n(\varphi)}{\partial \varphi} > 0.$ Second, for any given market j, we have already shown that bilateral export revenues and profits increase with management ability, $\frac{dr_j(\varphi)}{d\varphi} > 0$ and $\frac{d\pi_j(\varphi)}{d\varphi} > 0$. From the product expertise cut-off condition for exporting, we know that $\frac{d\lambda_j^*(\varphi)}{d\varphi} < 0$. This implies that a higher φ is associated with a bigger measure of products $N_j(\varphi) = 1 - Z\left(\lambda_j^*(\varphi)\right)$ exported to destination j:

$$\frac{dN_{j}(\varphi)}{d\varphi} = -\frac{dZ\left(\lambda_{j}^{*}(\varphi)\right)}{d\varphi} = -\frac{dZ\left(\lambda_{j}^{*}(\varphi)\right)}{d\lambda_{j}^{*}}\frac{d\lambda_{j}^{*}(\varphi)}{d\varphi} > 0.$$
(1.34)

Third, total export sales $r_X(\varphi)$, profits $\pi_X(\varphi)$ and number of products $N_X(\varphi)$ are:

$$r_X(\varphi) = \sum_j r_j(\varphi) I\left(\varphi \ge \varphi_j^*\right), \quad \pi_X(\varphi) = \sum_j \pi_j(\varphi) I\left(\varphi \ge \varphi_j^*\right), \quad N_X(\varphi) = 1 - Z\left(\lambda_X^*(\varphi)\right)$$
(1.35)

where $\lambda_X^*(\varphi) = \min \{\lambda_1^*(\varphi), \lambda_2^*(\varphi), ..., \lambda_J^*(\varphi)\}$ denotes the minimum product expertise cut-off for exporting $\lambda_j^*(\varphi)$ across countries j for a firm with given φ . Note that firms export a nested set of products i to different markets, which follows a strict pecking order based on λ_i . Since $\frac{dr_j(\varphi)}{d\varphi} > 0$, $\frac{d\pi_j(\varphi)}{d\varphi} > 0$, $\frac{\partial I(\varphi \ge \varphi_j^*)}{\partial \varphi} > 0$ and $\frac{dN_j(\varphi)}{d\varphi} > 0$, it directly follows that:

$$\frac{dr_X(\varphi)}{d\varphi} = \sum_j \left[\frac{dr_j(\varphi)}{d\varphi} I\left(\varphi \ge \varphi_j^*\right) + \frac{dI\left(\varphi \ge \varphi_j^*\right)}{d\varphi} r_j(\varphi) \right] > 0, \quad (1.36)$$

$$\frac{d\pi_X(\varphi)}{d\varphi} = \sum_j \left[\frac{d\pi_j(\varphi)}{d\varphi} I\left(\varphi \ge \varphi_j^*\right) + \frac{dI\left(\varphi \ge \varphi_j^*\right)}{d\varphi} \pi_j(\varphi) \right] > 0, \quad (1.37)$$

$$\frac{dN_X(\varphi)}{d\varphi} = -\frac{dZ(\lambda_X^*(\varphi))}{d\lambda_X^*}\frac{d\lambda_X^*(\varphi)}{d\varphi} > 0.$$
(1.38)

Proposition 3. Better managed firms offer higher-quality products if $\theta > 0$, but quality is invariant across firms if $\theta = 0$. Better managed firms set lower quality-adjusted prices if $\delta > 0$, but quality-adjusted prices are invariant across firms if $\delta = 0$. Better managed firms charge higher prices if $\theta > \delta$ and lower prices if $\delta > \theta$, but prices are invariant across firms if $\theta = \delta$.

Proof. This proposition can be established directly from the solution to the firm's profitmaximization problem above. Taking the partial derivative of firm's price, quality and quality-adjusted price with respect to management ability, we have:

$$p_{ji}(\varphi,\lambda_i) = \frac{\tau_j(\varphi\lambda_i)^{\theta-\delta}}{\alpha} \Longrightarrow \frac{\partial p_{ji}}{\partial \varphi} = \frac{(\theta-\delta)}{\alpha} \tau_j(\varphi\lambda_i)^{\theta-\delta-1} \lambda_i$$
(1.39)

$$q_{ji}(\varphi,\lambda_i) = (\varphi\lambda_i)^{\theta} \Longrightarrow \frac{\partial q_{ji}}{\partial \varphi} = \theta (\varphi\lambda_i)^{\theta-1}\lambda_i$$
(1.40)

$$\frac{p_{ji}(\varphi,\lambda_i)}{q_{ji}(\varphi,\lambda_i)} = \frac{\tau_j(\varphi\lambda_i)^{-\delta}}{\alpha} \Longrightarrow \frac{\partial (p_{ji}/q_{ji})}{\partial \varphi} = -\frac{\delta}{\alpha} \tau_j(\varphi\lambda_i)^{-\delta-1} \lambda_i$$
(1.41)

Recall that $\theta \ge 0$ and $\delta \ge 0$. It immediately follows that $\frac{\partial q_{ji}}{\partial \varphi} > 0$ if and only if $\theta > 0$ and $\frac{\partial (p_{ji}/q_{ji})}{\partial \varphi} < 0$ if and only if $\delta > 0$. Since the sign of $\frac{\partial p_{ji}}{\partial \varphi}$ depends on $(\theta - \delta)$, $\frac{\partial p_{ji}}{\partial \varphi} > 0$ if $\theta > \delta$, $\frac{\partial p_{ji}}{\partial \varphi} < 0$ if $\delta > \theta$, and $\frac{\partial p_{ji}}{\partial \varphi} = 0$ if $\theta = \delta$.

Proposition 4. Better managed firms use more expensive inputs of higher quality and/or more expensive assembly of higher complexity if $\theta > 0$, but input quality and assembly complexity are invariant across firms if $\theta = 0$.

Proof. From Proposition 3, we know that better managed firms produce goods of higher quality if and only if $\theta > 0$. While we do not explicitly model firms' endogenous choice of product quality in the baseline framework, we assume that producing goods of higher quality entails higher marginal production costs. The implicit micro-foundation for this quality production function is that manufacturing higher-quality products requires more expensive inputs of higher quality and/or more costly assembly technologies. See also Section 2.3 in this Appendix.

2 Model Extensions

2.1 Extension 1: Endogenous Management

Our baseline model assumes that management competence is an exogenous draw at the firm level. We now establish that Propositions 1-4 would continue to hold if an exogenous firm primitive endogenously determines the firm's choice of management practice, as long as implementing more effective management practices improves firm performance but is sufficiently more costly. Intuitively, adopting more sophisticated management practices can enhance existing firm capabilities and thereby stimulate market entry and firm revenues. Good management and intrinsic firm attributes may also be complementary, such that effective firm productivity may be supermodular in these two components. At the same time, superior management strategies arguably require higher sunk costs of adoption (e.g. hiring a manager, re-designing production facilities, training staff to use new data monitoring, etc.) and higher fixed costs of production (e.g. collecting data, analyzing peformance, communicating results to staff, etc.). As a result of such economies of scale, exogenously better firms that expect to be more competitive in the market and generate higher sales would endogenously choose better management practices, thereby further improving their performance. Propositions 1-4 would then hold both for the exogenous firm primitive and for the endogenous management quality. In particular, the Propositions would state causal effects for the firm primitive and conditional correlations for management, where the latter would constitute one mechanism through which the former operates.

To illustrate this insight tractably and transparently, we make minimal functional form assumptions for the impact of management choice on firm ability and for the cost of management adoption. The same insight would however apply more generally, as long as the benefit to management upgrading increases faster with management competence than the cost of management upgrading.

We assume that firm entrepreneurs receive an exogenous talent draw ϕ and choose to use management practice m at a convex fixed cost of f_m , where $df_m/dm > 0$ and $d^2 f_m/dm^2 > 0$. Firm ability $\varphi = \phi m(\phi)$ depends on the combination of talented entrepreneurs and management effectiveness. Given product expertise draws λ_i , firms can produce one unit of product i with quality $q_i = [\varphi \lambda_i]^{\theta} = [\phi m(\phi) \lambda_i]^{\theta}$ at a marginal cost of $[\varphi \lambda_i]^{\theta-\delta} = [\phi m(\phi) \lambda_i]^{\theta-\delta}$. In this environment, the proof below establishes that Propositions 1-4 continue to hold as conditional correlations for the endgenous management level in two steps: We first show that Propositions 1-4 apply for effective firm ability $\varphi = \phi m(\phi)$. We then demonstrate that effective firm ability and management are monotonically related, $d\varphi/dm > 0$. Together, these two results directly imply that Propositions 1-4 must also hold for management $m(\phi)$.

Proof. Step One

This extension of the model closely follows the solution concept in Sections 1.3 and 1.4 of this Appendix. Since the fixed cost of management adoption is independent of the firm's product scope, market penetration, and production scale, the firms' profit maximization problem can be solved in steps. The choice of management practice will be determined in

the last of these steps. All preceding steps will remain in essense the same as in the baseline model, such that all key equations can be obtained simply by replacing φ with $\phi m(\phi)$.

First, note that entrepreneurial talent ϕ and management competence m always enter multiplicatively as firm ability $\varphi = \phi m(\phi)$ and fix product quality at $q_i = [\varphi \lambda_i]^{\theta} = [\phi m(\phi) \lambda_i]^{\theta}$. The firm will therefore begin by choosing the profit-maximizing price and quantity in each potential destination-product market, conditional on entry there. The optimal price, quantity, quality-adjusted price, revenues and profits for product i in country jwill be given by equations (1.17). In particular, domestic profits $\pi_{di}(\phi, m, \lambda)$ from product iand export profits $\pi_{ji}(\phi, m, \lambda_i)$ from product i in country j will be given by the expressions below and increasing in management competence as before:

$$\pi_{di}(\phi, m, \lambda_i) = \frac{1}{\sigma} R_d \left(P_d \alpha \right)^{\sigma-1} \left(\phi m \lambda_i \right)^{\delta(\sigma-1)} - f_p \Longrightarrow \frac{\partial \pi_{di}(\phi, m, \lambda)}{\partial m} > 0$$
(2.1)

$$\pi_{ji}(\phi, m, \lambda_i) = \frac{1}{\sigma} R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1} (\phi m \lambda_i)^{\delta(\sigma-1)} - f_{pj} \Longrightarrow \frac{\partial \pi_{ji}(\phi, m, \lambda)}{\partial m} > 0 \quad (2.2)$$

Second, the firm will decide which products to produce and which products to export to destination j based on product expertise cut-offs for production and for exporting, $\lambda^*(\phi, m)$ and $\lambda_j^*(\phi, m)$. As before, these cut-offs are given by zero-profit conditions and defined by equations (1.23) and (1.25). However, these are no longer closed-form solutions that depend only on the exogenous firm attribute φ and model parameters, since firm ability $\varphi = \phi m(\phi)$ is now endogenous. Note also that these product expertise cut-offs are decreasing in both entrepreneurial talent and management capacity:

$$\lambda^{*}(\phi,m) = \frac{1}{\phi m} \left[\frac{\sigma f_{p}}{R_{d} \left(P_{d} \alpha \right)^{\sigma-1}} \right]^{\frac{1}{\delta(\sigma-1)}}, \frac{\partial \lambda^{*}(\phi,m)}{\partial m} = -\frac{\lambda^{*}}{m} < 0, \ \frac{\partial \lambda^{*}(\phi,m)}{\partial \phi} = -\frac{\lambda^{*}}{\phi} < 0 \ (2.3)$$
$$\lambda^{*}_{j}(\phi,m) = \frac{1}{\phi m} \left[\frac{\sigma f_{pj}}{R_{j} \left(\frac{P_{j} \alpha}{\tau_{j}} \right)^{\sigma-1}} \right]^{\frac{1}{\delta(\sigma-1)}}, \frac{\partial \lambda^{*}_{j}(\phi,m)}{\partial m} = -\frac{\lambda^{*}_{j}}{m} < 0, \ \frac{\partial \lambda^{*}_{j}(\phi,m)}{\partial \phi} = -\frac{\lambda^{*}_{j}}{\phi} < 0 \ (2.4)$$

Third, the firm will choose which export markets
$$j$$
 to enter. This decision will be guided
by firm ability cut-offs for exporting, φ_j^* , which are pinned down by the zero-profit condition

 $\pi_j(\varphi_j^*) = 0$ as earlier. Together, the results above imply that Propositions 1-4 hold for effective firm ability

Fogether, the results above imply that Propositions 1-4 hold for elective initial $\varphi = \phi m(\phi)$.

Step Two

Given Step One above, Propositions 1-4 will atutomatically hold for management competence m if effective firm ability $\varphi = \phi m(\phi)$ is increasing in m. We now prove this monotonicity.

In the final stage of the firm's problem, the entrepreneur will decide whether to begin production upon learning his talent draw. It is at this point that the firm will also choose its optimal management practice m and thereby effective ability φ , in order to maximize global profits from domestic sales and any exports abroad. This profit maximization problem closely resembles equation (1.29) in the baseline model:

$$\max_{m} \pi\left(\phi, m\right) = \int_{\lambda^{*}(\phi, m)}^{\infty} \pi_{di}\left(\phi, m, \lambda\right) z\left(\lambda\right) d\lambda + \sum_{j} \left(\int_{\lambda^{*}_{j}(\phi, m)}^{\infty} \pi_{ji}\left(\phi, m, \lambda\right) z\left(\lambda\right) d\lambda - f_{hj}\right) - f_{h} - f_{m} d\lambda$$

$$(2.5)$$

The first order condition with respect to management practices m implies that:

$$\frac{\partial \pi (\phi, m)}{\partial m} = \left(\int_{\lambda^*(\phi, m)}^{\infty} \frac{\partial \pi_{di} (\phi, m, \lambda)}{\partial m} z(\lambda) d\lambda - \pi_{di} (\phi, m, \lambda^*) z(\lambda^*) \frac{\partial \lambda^* (\phi, m)}{\partial m} \right) +$$

$$+ \sum_{j} \left(\int_{\lambda^*_{j}(\phi, m)}^{\infty} \frac{\partial \pi_{ji} (\phi, m, \lambda)}{\partial m} z(\lambda) d\lambda - \pi_{ji} (\phi, m, \lambda^*_{j}) z(\lambda^*_{j}) \frac{\partial \lambda^*_{j} (\phi, m)}{\partial m} \right) - \frac{\partial f_m}{\partial m} =$$

$$= \int_{\lambda^*(\phi, m)}^{\infty} A_d \delta (\sigma - 1) \frac{(\phi m \lambda)^{\delta(\sigma - 1)}}{m} z(\lambda) d\lambda +$$

$$+ \sum_{j} \left(\int_{\lambda^*_{j}(\phi, m)}^{\infty} A_{j} \delta (\sigma - 1) \frac{(\phi m \lambda)^{\delta(\sigma - 1)}}{m} z(\lambda) d\lambda \right) - \frac{\partial f_m}{m}$$

$$= 0.$$
(2.6)

Note that by the definition of the zero-profit product expertise cut-offs $\lambda^*(\phi, m)$ and $\lambda_j^*(\phi, m)$, the terms involving $\pi_{di}(\phi, m, \lambda^*) = \pi_{ji}(\phi, m, \lambda_j^*) = 0$ drop out. For ease of notation, the exogenous terms characterizing aggregate expenditure, aggregate price indices, and bilateral trade costs have been collected in $A_d \triangleq \frac{1}{\sigma} R_d (P_d \alpha)^{\sigma-1}$ and $A_j \triangleq \frac{1}{\sigma} R_j \left(\frac{P_j \alpha}{\tau_j}\right)^{\sigma-1}$.

Using this first order condition, one can solve for the firm's optimal management competence level m as an implicit function of ϕ defined as $F(\phi, m)$:

$$F(\phi,m) \equiv \int_{\lambda^{*}(\phi,m)}^{\infty} A_{d}\delta(\sigma-1) \frac{(\phi m\lambda)^{\delta(\sigma-1)}}{m} z(\lambda) d\lambda +$$

$$+ \sum_{j} \left(\int_{\lambda^{*}_{j}(\phi,m)}^{\infty} A_{j}\delta(\sigma-1) \frac{(\phi m\lambda)^{\delta(\sigma-1)}}{m} z(\lambda) d\lambda \right) - \frac{\partial f_{m}}{m}$$

$$\triangleq F_{d}(\phi,m) + \sum_{j} F_{j}(\phi,m) - \frac{\partial f_{m}}{m}.$$

$$(2.7)$$

We want to prove that $\varphi = \phi m(\phi)$ is increasing in *m*. We therefore need to show that:

$$\frac{d\left(\phi m\left(\phi\right)\right)}{dm} = \frac{d\phi}{dm}m + \phi = \phi\left(\frac{d\phi}{dm}\frac{m}{\phi} + 1\right) > 0.$$
(2.8)

From the Implicit Function Theorem, it follows that:

$$\frac{d\phi}{dm} = -\frac{\partial F/\partial m}{\partial F/\partial \phi}.$$
(2.9)

Therefore, all we need is to prove that:

$$\frac{\partial F/\partial m}{\partial F/\partial \phi} < \frac{\phi}{m}.$$
(2.10)

We first show that the denominator $\partial F/\partial \phi$ is positive. Note that

$$\frac{\partial F}{\partial \phi} = F_{1d}(\phi, m) + \sum_{j} F_{1j}(\phi, m), \qquad (2.11)$$

where for each country k in the set comprising the home economy d and all potential export destinations $j, k \in \{d\} \cup \{1, 2, ..., J\}, F_{1k}(\phi, m)$ is given by:

$$F_{1k}(\phi,m) = A_k \delta(\sigma-1) (\phi m)^{\delta(\sigma-1)-1} \left[\delta(\sigma-1) \int_{\lambda_k^*(\phi,m)}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda) \, d\lambda - \phi(\lambda_k^*)^{\delta(\sigma-1)} \, z(\lambda_k^*) \, \frac{\partial \lambda_k^*(\phi,m)}{\partial \phi} \right]$$
(2.12)

Since $\partial \lambda_k^*(\phi, m) / \partial \phi < 0$ as shown above, it follows that $\partial F / \partial \phi > 0$. We next examine the numerator $\partial F / \partial m$:

$$\frac{\partial F}{\partial m} = F_{2d}\left(\phi, m\right) + \sum_{j} F_{2j}\left(\phi, m\right) - \frac{d^2 f_m}{dm^2},\tag{2.13}$$

where for each country k, $F_{2k}(\phi, m)$ is given by:

$$F_{2k}(\phi,m) = A_k \delta(\sigma-1) (\phi m)^{\delta(\sigma-1)} \frac{1}{m} \left[\frac{\delta(\sigma-1)-1}{m} \int_{\lambda_k^*}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda) \, d\lambda - (\lambda_k^*)^{\delta(\sigma-1)} z(\lambda_k^*) \frac{\partial \lambda_k^*}{\partial m} \right]$$
(2.14)

Since $\partial F/\partial \phi > 0$ and $d^2 f_m/dm^2 > 0$, we therefore know that:

$$\frac{\partial F/\partial m}{\partial F/\partial \phi} < \frac{F_{2d}\left(\phi,m\right) + \sum_{j} F_{2j}\left(\phi,m\right)}{F_{1d}\left(\phi,m\right) + \sum_{j} F_{1j}\left(\phi,m\right)}.$$
(2.15)

Recalling that $\partial \lambda_k^* / \partial \phi = -\lambda_k^* / \phi$ and $\partial \lambda_k^* / \partial m = -\lambda_k^* / m$ for all $k \in \{d\} \cup \{1, 2, \dots, J\}$, one can show that $F_{2k}(\phi, m) / F_{1k}(\phi, m) < \phi / m$:

$$\frac{F_{2k}(\phi,m)}{F_{1k}(\phi,m)} = \frac{A_k \delta(\sigma-1)(\phi m)^{\delta(\sigma-1)} \frac{1}{m} \left[\frac{\delta(\sigma-1)-1}{m} \int_{\lambda_k^*}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda) d\lambda - (\lambda_k^*)^{\delta(\sigma-1)} z(\lambda_k^*) \frac{\partial \lambda_k^*}{\partial m} \right]}{A_k \delta(\sigma-1)(\phi m)^{\delta(\sigma-1)-1} \left[\delta(\sigma-1) \int_{\lambda_k^*}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda) d\lambda - \phi(\lambda_k^*)^{\delta(\sigma-1)} z(\lambda_k^*) \frac{\partial \lambda_k^*}{\partial \phi} \right]}$$

$$(2.16)$$

$$= \frac{\phi \left[\frac{\delta(\sigma-1)-1}{m} \int_{\lambda_k^*}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda) d\lambda + (\lambda_k^*)^{\delta(\sigma-1)} z(\lambda^*) \frac{\lambda_k^*}{m} \right]}{\left[\delta(\sigma-1) \int_{\lambda_k^*}^{\infty} \lambda^{\delta(\sigma-1)} z(\lambda^*) d\lambda + \phi(\lambda_k^*)^{\delta(\sigma-1)} z(\lambda^*) \frac{\lambda_k^*}{m} \right]}$$

$$= \frac{\phi}{m} \cdot \frac{\left[\delta\left(\sigma-1\right)\int_{\lambda_{k}^{*}}^{\infty} \lambda^{\delta\left(\sigma-1\right)} z\left(\lambda\right) d\lambda + \phi\left(\lambda_{k}^{*}\right)^{\delta\left(\sigma-1\right)} z\left(\lambda_{k}^{*}\right) \frac{\lambda_{k}^{*}}{\phi}\right]}{\delta\left(\sigma-1\right) - 1\right)\int_{\lambda_{k}^{*}}^{\infty} \lambda^{\delta\left(\sigma-1\right)} z\left(\lambda\right) d\lambda + \left(\lambda_{k}^{*}\right)^{\delta\left(\sigma-1\right)} z\left(\lambda_{k}^{*}\right) \lambda_{k}^{*}}}{\delta\left(\sigma-1\right)\int_{\lambda_{k}^{*}}^{\infty} \lambda^{\delta\left(\sigma-1\right)} z\left(\lambda\right) d\lambda + \left(\lambda_{k}^{*}\right)^{\delta\left(\sigma-1\right)} z\left(\lambda_{k}^{*}\right) \lambda_{k}^{*}}} < \frac{\phi}{m}.$$

Therefore,

$$\frac{\partial F/\partial m}{\partial F/\partial \phi} < \frac{F_{2d}\left(\phi,m\right) + \sum_{j} F_{2j}\left(\phi,m\right)}{F_{1d}\left(\phi,m\right) + \sum_{j} F_{1j}\left(\phi,m\right)} < \frac{\phi}{m}.$$
(2.17)

We have thus proven that effective firm ability $\varphi = \phi m(\phi)$ is increasing in management competence *m*. Since all comparative statics for φ hold as in the baseline model, it follows that all propositions also hold as conditional correlations for management quality *m* even when firms endogenously choose their management practices.

2.2 Extension 2: Multiple Ability Components

The theoretical predictions of our baseline model would continue to hold if management is one of multiple draws that jointly determine firm ability φ . For example, firm ability $\varphi = m \cdot \phi$ may depend on the entrepreneur's intrinsic talent ϕ and the manager's competence for implementing effective management practices m. If entrepreneurs and managers do not match perfectly assortatively due to labor market frictions, then $|corr(m, \phi)| \neq 1$. While all firm outcomes would now be pinned down by φ instead of m alone, management competence would have the same effects as in our baseline model *ceteris paribus*. Propositions 1-4 would now hold for φ unconditionally, for ϕ conditional on m, and for m conditional on ϕ . The last result is the conditional relationship that remains relevant for our empirical analysis.

2.3 Extension 3: Endogenous Quality

For expositional simplicity, we do not model firms's choice of product quality in the baseline model, and adopt instead a reduced-form quality production function. Endogenizing firms' choice of input and output quality in a richer framework would however preserve our theoretical predictions. What is sufficient for this to occur is that output quality - and by extension firm profits - is supermodular in firm ability and either the quality of inputs or the complexity of the assembly process. We illustrate this point here by incorporating endogenous quality choice as in Kugler and Verhoogen (2012) into our baseline framework. The same key insights would emerge with alternative microfoundations for the quality production function.

We assume that there is complementarity between firm ability and input quality in the production of output quality. In particular, using an input of quality c_{ji} , the firm can produce one unit of product *i* with output quality

$$q_{ji} = \left[\frac{1}{2}\left(\left(\varphi\lambda_i\right)^b\right)^\rho + \frac{1}{2}\left(c_{ji}^2\right)^\rho\right]^{\frac{1}{\rho}}$$
(2.18)

at a marginal cost of c_{ji} . In this setting, the parameter b can be interpreted as the scope for quality differentiation, while the parameter ρ governs the degree of complementarity between input quality c_{ji} and firm-specific management φ (as well as firm-product specific expertise λ_i). The quadratic specification for c_{ji} is not crucial but adopted for tractability. Given this quality production function, more capable firms will optimally use higherquality inputs in order to produce higher-quality goods.

Proof. Now the firm's maximization problem becomes

$$\max_{p_{ji}, x_{ji}, c_{ji}} \pi\left(\varphi, \lambda_i\right) = p_{ji} x_{ji} - \tau_j x_{ji} c_{ji} - f_{pj}$$

s.t. $x_{ji} = R_j P_j^{\sigma-1} q_{ji}^{\sigma-1} p_{ji}^{-\sigma}$

Substituting the constraint into the objective function, this is equivalent to solving

$$\max_{p_{ji}, c_{ji}} \pi_{ji} \left(\varphi, \lambda_{i}\right) = R_{j} P_{j}^{\sigma-1} \left[\frac{1}{2} \left(\varphi\lambda_{i}\right)^{b\rho} + \frac{1}{2} c_{ji}^{2\rho}\right]^{\frac{\sigma-1}{\rho}} p_{ji}^{-\sigma} \left(p_{ji} - \tau_{j} c_{ji}\right) - f_{pj}$$

The first order conditions with respect to p_{ji} and c_{ji} yield the following equations respectively:

$$p_{ji} = \frac{\sigma}{\sigma - 1} \tau_j c_{ji} \tag{2.19}$$

$$(\sigma - 1) c_{ji}^{2\rho - 1} (p_{ji} - \tau_j c_{ji}) = \tau_j \left[\frac{1}{2} (\varphi \lambda_i)^{b\rho} + \frac{1}{2} c_{ji}^{2\rho} \right]$$
(2.20)

Substituting equation (2.19) into equation (2.20) and using equation (2.18) delivers the following endogenous input quality c_{ji} and output quality q_{ji} as a function of firm management ability φ and product expertise λ_i :

$$c_{ji} = c_i = (\varphi \lambda_i)^{\frac{b}{2}}, \quad q_{ji} = q_i = (\varphi \lambda_i)^b.$$
(2.21)

This expression immediately implies that better managed firms will endogenously choose to source higher-quality inputs in order to produce higher-quality goods, i.e. $\frac{\partial c_i(\varphi,\lambda_i)}{\partial \varphi} > 0$ and $\frac{\partial q_i(\varphi,\lambda_i)}{\partial \varphi} > 0$. While we have allowed firms to freely vary input and output quality across markets j, the quality production function we have considered guarantees that firms optimally select a single quality level for each product i in their portfolio. Intuitively, better managed firms would endogenously produce higher-quality goods for any given market under alternative formulations that allow for quality customization across markets.

Finally, note that when $\theta = b$ and $\delta = \frac{b}{2}$, the solution in equation (2.21) corresponds exactly to the reduced-form formulation of the quality production function in our baseline model: Firms then produce one unit of product *i* with quality $q_i = (\varphi \lambda_i)^{\theta}$ at a marginal cost of $c_i = (\varphi \lambda_i)^{\theta - \delta}$.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

1552	Gabriel M. Ahlfeldt Wolfgang Maennig Steffen Q. Mueller	The Generation Gap in Direct Democracy
1551	Keith Head Tierry Mayer	Brands in Motion: How Frictions Shape Multinational Production
1550	Natalie Chen Dennis Novy	Currency Unions, Trade and Heterogeneity
1549	Felipe Carozzi	The Role of Demand in Land Re-Development
1548	Nick Bloom Brian Lucking John Van Reenen	Have R&D Spillovers Changed?
1547	Laura Alfaro Nick Bloom Paola Conconi Harald Fadinger Patrick Legros Andrew F. Newman Raffaella Sadun John Van Reenen	Come Together: Firm Boundaries and Delegation
1546	Jörg Kalbfuß Reto Odermatt Alois Stutzer	Medical Marijuana Laws and Mental Health in the United States
1545	Maria Molina-Domene	Specialization Matters in the Firm Size-Wage Gap
1544	Jonathan Colmer	Weather, Labor Reallocation and Industrial Production: Evidence from India
1543	Andrew B. Bernard Esther Ann Bøler Swati Dhingra	Firm-to-Firm Connections in Colombian Imports

1542	Gerard H. Dericks Hans R.A. Koster	The Billion Pound Drop: The Blitz and Agglomeration Economics in London
1541	Andrew B. Bernard Andreas Moxnes	Networks and Trade
1540	Does Gentrification Displace Poor Households? An 'Identification-Via- Interaction' Approach	Sevrin Waights
1539	Zhuan Pei Jörn-Steffen Pischke Hannes Schwandt	Poorly Measured Confounders Are More Useful On the Left than On the Right
1538	Konstantin Büchel Stephan Kyburz	Fast Track to Growth? Railway Access, Population Growth and Local Displacement in 19 th Century Switzerland
1537	Stephen Gibbons Maria Sánchez-Vidal Olmo Silva	The Bedroom Tax
1536	Pawel Adrjan Brian Bell	Pension Shocks and Wages
1535	Tom Kirchmaier Stephen Machin Matteo Sandi Robert Witt	Prices, Policing and Policy: The Dynamics of Crime Booms and Busts
1534	Laurent Bouton Paola Conconi Francisco Pino Maurizio Zanardi	Guns, Environment and Abortion: How Single-Minded Voters Shape Politicians Decisions
1533	Giulia Giupponi Stephen Machin	Changing the Structure of Minimum Wages: Firm Adjustment and Wage Spillovers

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP_LSE